103 Main Street, Suite #3 Meredith, NH 03253 Tel 603.279.8171 www.lakesrpc.org



## LRPC EXECUTIVE BOARD MEETING Wednesday, June 12, 2024 9:00 AM **LRPC Office, First Floor Conference Room** Humiston Building, 103 Main Street, Meredith, NH AGENDA NOTE: Seven (7) Executive Board members must be present in-person at the meeting location to establish a quorum, so please let Linda know if you cannot attend. **Call to Order** Approve Draft Minutes of May 8, 2024 (actionable item) Attachment 1 Finance/Treasurer Report (May) Attachment 2 Monthly Executive Report (May) Attachment 3 **Committee Reports** May Transportation Advisory Committee (TAC) a. **Old Business** Staff Updates a. b. Annual Meeting Final Update **New Business** a. Bike/Ped Plan Adoption (actionable item) Attachment 4 b. Certificate of Authority (actionable item) Attachment 5 TYP Project Approval – Plymouth (actionable item) C. d. Area Commissioner Appointment (*potential actionable item*) Non-Public Executive Session (RSA 91-A:3, II(b) - Personnel) Roundtable Adjourn

### NEXT MEETING: September 11, 2024 **ENJOY YOUR SUMMER!**

The Lakes Region Planning Commission reserves the right to hold a non-public session whether noted on the Agenda or not. Notice of a nonpublic session on an agenda is for planning purposes only. The citations to the Right-to-Know Law are provisional and may be revised as circumstances required. The LRPC complies with the ADA regulations. Please contact the LRPC office if you need special assistance in order to attend this meeting.

All meeting dates and times are subject to change.

1.

2.

3.

4.

5.

6.

7.

8.

9.

10.

Attachment I

103 Main Street, Suite #3 Meredith, NH 03253 Tel 603.279.8171 www.lakesrpc.org



# LRPC Executive Board Meeting Minutes of May 8, 2024

PRESENT	Jane Alden, Dean Anson, Cristina Ashjian, Tavis Austin, John Ayer (Chair), Mardean Badger, Bill Bolton, Mark Hildebrand, David Katz (Vice Chair), Lianne Prentice (Secretary)
ABSENT	David Kerr (Treasurer), Robert Lambert, Stephanie Maltais
STAFF	Jeff Hayes (Executive Director), Linda Waldron (Executive Administrative Assistant)
LOCATION	LRPC Office, 1 <sup>st</sup> Floor Conference Room, Meredith

### 1. Call to Order

Chair Ayer called the meeting to order at 9:03 AM. A quorum was established.

### 2. Minutes

The minutes of April 10<sup>th</sup> were presented for approval. Vice Chair Katz made a motion to accept the minutes as submitted, seconded by Secretary Prentice. A voice vote was taken indicating all present were in favor, except Jane Alden who abstained. Motion passed.

Chair Ayer advised that there would be a change in the order of business on the Agenda as one member may need to depart early and would like to be present for the non-public session. Executive Director Hayes was asked to leave the room for the non-public session. Executive Director Hayes noted that he had a right to stay, but would not exercise that right. As such, Vice Chair Katz moved to go forward into non-public session which was seconded by Secretary Prentice. A roll call vote was taken with all in attendance voting in the affirmative. Motion passed and the non-public session began at 9:06 AM.

At 9:49 AM, Vice Chair Katz offered a motion to exit the non-public session and seal the minutes for a period of one year. This was seconded by Secretary Prentice. A roll call vote was taken with all in attendance voting in the affirmative. Motion passed and the non-public session ended. A brief recess was taken.

### 3. Finance/Treasurer Reports

Executive Director Hayes noted that finances continue to look good. There were some unexpected expenses such as the need to hire a recruiter to solicit applicants for the Executive Director opening, but we should end the year in a healthy position.

### 4. Monthly Executive Reports

Executive Director Hayes reviewed the report briefly. He stated that transportation is in a good place and it is anticipated that Sean Chamberlin's position should be filled by September. The next ten year plan round will be the first priority for his replacement.

Vice Chair Katz interjected and proffered a motion that the Executive Board recommend to the TAC that they suspend accepting any new applications to the ten year plan update and instead reevaluate prior submissions. Chair Ayer seconded the motion. A voice vote was taken indicating all present were in favor. Motion passed. Executive Director Hayes was asked to extend this recommendation to the TAC.

ALEXANDRIA • ANDOVER • ASHLAND • BARNSTEAD • BELMONT • BRIDGEWATER • BRISTOL • CENTER HARBOR • DANBURY EFFINGHAM • FRANKLIN • FREEDOM • GILFORD • GILMANTON • HEBRON • HILL • HOLDERNESS • LACONIA • MEREDITH • MOULTONBOROUGH NEW HAMPTON • NORTHFIELD • OSSIPEE • PLYMOUTH • SANBORNTON • SANDWICH • TAMWORTH • TILTON • TUFTONBORO • WOLFEBORO

103 Main Street, Suite #3 Meredith, NH 03253 Tel 603.279.8171 www.lakesrpc.org



### 5. Committee Reports

- a. <u>Transportation Advisory Committee (TAC)</u>. In the absence of Treasurer Kerr, Executive Director Hayes reported that there was a meeting of the TAC held on May 1<sup>st</sup> in Tuftonboro which was the final meeting of the fiscal year. The bike/ped plan should be ready for adoption by the Board for June's meeting. The regional transportation plan still needs to be finalized with missing crash data in order to make our municipalities eligible for direct transportation funding.
- b. <u>Annual Awards Nominees | Committee Recommendations</u>. Only two nominations were received: one for the Award of Excellence and one for the Community Service Award. Both are strong candidates, and both are recommended by the Committee to be recipients in their respective categories. There was some discussion about the lack of an Ayers-Cotton nominee. Vice Chair Katz made a motion to accept the Awards Committee recommendations as submitted, seconded by Secretary Prentice. A voice vote was taken indicating all present were in favor. Motion passed.

Based on last year's decision to recognize the service of members of the commission who have retired/resigned during the year, a review of candidates for FY24 was had which resulted in 6 recognition awards and one recognition certificate (<5 yrs. service = certificate; >5 yrs. service = award). Chair Ayer thanked the Awards Committee for their time and recommendations.

### 6. Old Business

a. <u>Annual Meeting Update</u>. Sponsors were reviewed with a total of \$3,000.00 having been received. It was noted that we only targeted prior sponsors this year rather than our large business database in order to simplify the process given that past sponsorship requests have yielded only the same core sponsors. A draft invitation was reviewed with no changes noted. It is anticipated that the invitation will go out next week along with the election ballot (ballot only distributed to the commissioners).

### 7. New Business

- a. <u>Annual Audit Contract</u>. Executive Director Hayes mentioned that this is the same firm we have been using for the past several years, but the Board has the option of choosing another firm should they wish. The proposed contract was reviewed, and it was determined to be satisfactory. Vice Chair Katz proposed a motion to authorize Executive Director Hayes to sign the contract on behalf of the Executive Board. Motion was seconded by Secretary Prentice. A voice vote was taken indicating all present were in favor. Motion passed.
- b/c. <u>DRIs</u>. Executive Director Hayes mentioned the projects listed on the agenda, and advised that our opinion letters as provided to the municipalities were available for review upon request.
- 8. Non-Public Executive Session hiring. This was addressed above.

### 9. Roundtable

D. Katz, New Hampton—Nothing significant to report.

L. Prentice, Tamworth—The transfer station is once again being reviewed for an upgrade as they lost the opportunity to purchase another piece of property to expand their recycling program.

M. Hildebrand, Center Harbor—The old Belknap College building and former Red Hill Inn property on Route 25B is being converted into an event facility. The Planning Board will be doing a walk through in a couple of weeks. Mark will be going to the Moultonborough Planning Board meeting tonight concerning the Harbor Landing project.

D. Anson, Laconia—Dean attended 2 meetings recently, both focused on water quality. The first was presented by the Lake Winnipesaukee Association concerning a grant study on Paugus Bay and the other by the Waukewan Watershed Advisory Committee; both water bodies are sources of drinking water. Both reports will be presented at the upcoming annual Drinking Water Source Protection Conference (May 16<sup>th</sup>).

103 Main Street, Suite #3 Meredith, NH 03253 Tel 603.279.8171 www.lakesrpc.org



J. Alden, Tilton—Tilton has an opening for a town administrator. The more recent TA had only served a couple of months before giving his notice. There is a group of people who have been coming in with an interest in subdividing some property, but they keep running into problems so not much progress has been made. There also is a developer looking to add 40 more townhouses to an existing development on Route 3.

B. Bolton, Plymouth—The water/sewer project on Route 3 was completed last fall and they will now be repaying the whole road (Route 3/Main Street) towards the end of next month. The Highland Street project is next, which includes installation of an oval rotary at the high school. The Smith Bridge Road traffic issue is on the ten year plan and scheduled for engineering in 2030. Bill noted that construction of a hotel planned in that same area will also add to the traffic problems.

M. Badger, Ashland—The golf course saga continues as they look to expand/improve parking which involves an easement on an adjacent property owned by the town. The Oxbow area of the Pemi River encompasses several communities who own parcels of land in that area and the town of Holderness is looking to purchase several of those pieces to preserve for conservation. Boundary surveys will ensue, and possible adjustments may be needed.

C. Ashjian, Moultonborough—Elections are next Tuesday with town meeting next Thursday. There is a warrant article proposal to move the annual town meeting back. Cristina toured the solar facility on Moultonborough Neck and attended the NH Preservation Alliance awards ceremony where Tilton and Ossipee received awards.

J. Ayer, Gilford—There was a proposal for the construction of dwelling units and the refurbishing of the inn on the old Kings Grant property but, unfortunately, that fell through and the property went up for auction. The town is waiting to see what will be proposed now. The substantial mountain property development project planned on the Alton/Gilford town line does not seem to be a concern for Gilford at this time. The proposal is for a high-end retreat and wedding venue with guest housing and not a concert venue with helicopter transportation as previously rumored. The Starbucks coffee shop has been nixed.

### 10. Adjournment

Chair Ayer adjourned the meeting at 11:30 AM.

Respectfully Submitted, Linda Waldron Executive Administrative Assistant

Attachment 2



# **Lakes Region Planning Commission - Dashboard**

### 1:12 PM 06/03/24 Accrual Basis

# **Lakes Region Planning Commission** STATEMENT OF FINANCIAL POSITION

As of May 31, 2024

	May 31, 24
ASSETS	
Current Assets	
Checking/Savings	
1011.00 · Operating Account	11,109.17
1040.00 · Petty Cash	150.00
1070.01 · Money Market	115,406.84
1071.00 · Savings	1,601.94
Total Checking/Savings	128,267.95
Accounts Receivable	
1110.00 · Accounts Receivable	195,443.22
Total Accounts Receivable	195,443.22
Other Current Assets	
1451.00 · Prepaid Postage	61.41
1452.00 · Bulk Mail Permit 98	30.46
1510.00 · Vanguard Investments	306,430.32
1580.00 · NH PDIP	198,610.14
Total Other Current Assets	505,132.33
Total Current Assets	828,843.50
Fixed Assets	0.00
TOTAL ASSETS	828,843.50
LIABILITIES & EQUITY Liabilities	
Current Liabilities	
Accounts Payable	
2010.00 · Accounts Payable	3,754.11
Total Accounts Payable	3,754.11
Credit Cards	1,504.08
Other Current Liabilities	
2110 00 · Accrued Salaries & Wages	838.00
2120.00 · Accrued Vacation Payroll	10.926.56
2130.00 · Payroll Liabilities	1,578.26
2350.00 · Deferred and Unearned Revenue	285,243.01
Total Other Current Liabilities	298,585.83
Total Current Liabilities	303,844.02
Total Liabilities	303,844.02
Equity	
3110.00 · Unrestricted Net Assets	555 186 70
Net Income	(30,187.22)
Total Equity	524,999.48
TOTAL LIABILITIES & EQUITY	828,843.50

-	-	4	-	-	-	_
the in sta		72.7	4		(in-ex.) 7/2	-
4	111		iii	-	-	-
			107	-		~
		к		P	- (	

# Lakes Region Planning Commission FY24 Budget Performance

May 31, 2024

			10	riscal tear	(24)-	to-Date	
	A Ann	PY24 pproved ual Budget	]	FY24 YTD Actual	l (Bud	Difference dget v Actual)	% of Annual Budget
Income							
4010 · Contract Income	\$	694,886	\$	488,380	\$	(206,506)	70.28%
4020 · HHW Income	\$	115,148	\$	110,487	\$	(4,661)	95.95%
4040 Interest Income	\$	6,163	\$	15,683	\$	9,520	254.46%
4050 Miscellaneous Income	\$	6,000	\$	6,950	\$	950	115.83%
4060 Local Appropriations	\$	141,704	\$	129,898	\$	(11,806)	91.67%
4070 · Sales Income	\$	1,300	\$	4,289	\$	2,989	329.92%
4015 Donations	\$	-	\$	200	\$	200	
4090 · Funds Transfer	\$	-	\$		\$	-	
Total income (est.)	\$	965,201	\$	755,886	\$	(209,315)	78.31%
Expense			-			Target:	91.67%
6030 · Custodian	\$	4,420	\$	3,995	\$	425	90.38%
6050 · Education and Training	\$	1,000	\$	4,588	\$	(3.588)	458.83%
6060 · Equipment Maintenance	\$	14,350	\$	13,767	\$	583	95.93%
6070 HHW Expense	\$	99,662	\$	96,678	\$	2,984	97.01%
6080 Insurance - Bonds and Business	\$	3,450	\$	3.469	\$	(19)	100.55%
7000 · Executive Director Recruitment	\$	-	\$	4.522	\$	(4,522)	
7010 · Publishing/Memberships/Meetings	\$	14,750	\$	8,716	\$	6.034	59.09%
7020 Miscellaneous Expense	\$	750	\$	1,423	\$	(673)	189.76%
7030 · Office Improvements	\$	1.000	\$	-	\$	1.000	0.00%
7040 Office Expense	\$	15,580	\$	14,168	\$	1,412	90,94%
7050 · Payroll Expenses	\$	717,933	\$	625,501	\$	92,432	87.13%
7060 · Postage and Printing	\$	1,050	\$	391	\$	659	37.21%
7069 · Allowance for Direct Grant Expenses	\$	-	\$	-	\$	-	
7070 · Professional Services	\$	48,220	\$	10.536	\$	37.684	21.85%
7080 · Rent	\$	9,672	\$	8,866	\$	806	91.67%
7090 · Traffic Equipment	\$	8,000	\$	563	\$	7,437	
7095 · SADES Equipment	\$	-	\$	-	\$	-	
8010 Travel Expense	\$	4,500	\$	3,359	\$	1,141	74.64%
8050 · Utilities	\$	15,364	\$	14,373	\$	991	93.55%
8060 · Vehicle O&M	\$	5,500	\$	3,606	\$	1,894	65.56%
Total Expense	\$	965,201	\$	818,520	\$	146,681	84.80%
	_					Target:	91.67%
Net Ordinary Income <sup>1</sup>	\$	(0)	\$	(62,634)			
Realized Gain (Loss) <sup>2</sup> Unrealized Gain (Loss) <sup>2</sup>			\$	32,447			

<sup>1</sup>Net Ordinary (Operating) Income is as of May 31, 2024. <sup>2</sup>Realized and Unrealized Investment Gain (Loss) are as of May 31, 2024.

00	1000		-
. Ш	III III		
A	111 111		
1111		TO	6

# Lakes Region Planning Commission FY24 Budget Performance May 31, 2024

				Fiscal Year			
	A Ann	FY24 Approved aual Budget		FY24 YTD Actual	(Bı	Difference udget v Actual)	% of Annual Budget
Membership		141,704		129,898		(11,806)	91.67%
Local Contracts							
PRLAC	\$	2,700	\$	4,200	\$	1,500	155.56%
DES   SWP (Belmont-Northfield-Sandwich)	\$	24,825	\$	16,977	\$	(7,849)	68.38%
RSMS Project   New Hampton Update	\$	1,298	\$	1,298	\$	-	100.00%
Gilford   CCDS			\$	11,123			
Other Traffic Counts			\$	540	\$	540	
NBRC   Gale School			\$	1,189	\$	1,189	
NBRC   Town of Sandwich			\$	2,363	\$	2,363	
NBRC   Town of Hebron			\$	1,952			
CDBG   Grafton County Micro Enterprise	\$	45,333	\$	51,708	\$	6,376	114.06%
NBRC   GALA Community Center			\$	1,776			
Additional NBRCs					\$	-	
Subtotal	\$	74,156	\$	93,126	\$	4,120	125.58%
State / Federal Contracts							
OEP Targeted Block Grant	\$	11,111	\$	10,034	\$	(1,077)	90.30%
PRLAC Corridor Management Plan	\$	1,700	\$	1,700	\$	-	100.00%
DOS   HSEM - Gilmanton	\$	7,500	\$	1,500	\$	(6,000)	20.00%
DOS   HSEM - Moultonborough	\$	7,500	\$	1,500	\$	(6,000)	20.00%
DOS   HSEM - Wolfeboro	\$	9,000	\$	1,800	\$	(7,200)	20.00%
DOS   HSEM - Andover	\$	7,500	\$	1,500	\$	(6,000)	20.00%
DOS-HSEM - Bridgewater	\$	1,500	\$	1,500	\$	-	100.00%
DOS-HSEM - Tilton					\$		
USDA   Solid Waste Management FY23	\$	39,063	\$	39,063	\$	0	100.00%
EPA   FY24/25 Brownfields	\$	72,133	\$	18,274	\$	(53,859)	25.33%
USDA   Solid Waste Management FY24	\$	106,500	\$	63,725	\$	(42,775)	59.84%
DOT UPWP FY 24 / 25	\$	305,411	\$	251,162	\$	(54,249)	82.24%
DOS   HSEM - Gilford							
DOT   HSIP Data Collection					\$		
HUD   Region Plan (Sheeheen Ear Mark)	\$	41,812	\$	3,495	\$	(38,317)	8.36%
NBRC   Admin and Coaching	\$	10,000			\$	(10,000)	
Subtotal	\$	620,730	\$	395,253	\$	(225,477)	63.68%
Total 4010 · Contract Income	\$	694,886	\$	488,380	\$	(221,357)	70.28%
Other Income							
4020 · HHW	\$	115,148	\$	110,487	\$	(4,661)	95.95%
4021 · HHW Local	\$	99.662	\$	95.677	\$	(3.985)	96.00%
4022 · DES HHW	\$	15 486	\$	14 810	.5	(676)	95.63%
4015 · Donations	\$	-	\$	200	S	200	00.0070
4040 · Interest & Dividends	S	6 163	\$	15 683	\$	9 520	254 46%
Land use book sales / GIS	\$	1 300	\$	4 289	\$	2 989	329 92%
Other/Misc Income/Annual Meeting	\$	6,000	\$	6,950	\$	950	115.83%
Fund Balance	\$	-	Ψ	0,000	\$	-	110.0070
Subtotal	\$	128,611	\$	137,608	\$	8,997	107 00%
		-,	+	,,	Ŧ	-,	
TOTAL	\$	965,201	\$	755,886	\$	(224,166)	78.31%
	-					Target:	91.67%

----



# Lakes Region Planning Commission FY24 Budget Performance May 31, 2024

	FY24 Approved Annual Budget		w	<b>Fiscal Year</b>			
			FY24 YTD Actual		Difference (Budget v Actual)		% of Annual Budget
ense Account	-						
6030 · Custodian	\$	1 120	¢	3 995	\$	125	00 38%
6050 · Education and Training	\$	1,420	¢ ¢	1 588	¢ ¢	(3 588)	458 839
6060 · Equipment Maintenance	÷	14 350	¢ ¢	13 767	Ψ ¢	(5,500)	450.057
6062 · Equipment/Computer Maintenance	¢ ¢	14,330	ф Ф	12 594	ф С	766	95.93
6061 · Equipment Purchases	¢ ¢	14,300	¢	13,004	\$	(193)	94.007
6070 - HUW Exponso	\$	00.662	\$ •	103	\$ •	(103)	07.040
6080 - Insurance Bonds and Business	\$	99,002	ф Ф	90,070	\$ \$	2,964	97.017
7000 - Executive Director Recruitment	Þ	3,450	¢	3,409	\$	(19)	100.55%
7000 · Executive Director Recruitment			\$	4,522	0	(1 500)	
7001 · Executive Search Fees			\$	4,500	\$	(4,500)	
7005 · Meetings	•		\$	22	\$	(22)	
7010 · Publishing/Wemberships/Meetings	\$	14,750	\$	8,716	\$	6,034	59.09
7020 · Miscellaneous Expense	\$	750	\$	1,423	\$	(673)	189.76
7030 · Office Improvements	\$	1,000	\$	-	\$	1,000	0.00
7040 · Office Expense	\$	15,580	\$	14,168	\$	1,412	90.94
7043 · Copier Lease	\$	5,000	\$	4,221	\$	779	84.419
7041 · Supplies	\$	10,580	\$	9,948	\$	632	94.029
7050 · Payroll Expenses	\$	717,933	\$	625,501	\$	92,432	87.139
7051 · Salaries and Wages	\$	526,833	\$	459,068	\$	67,765	87.149
7052 · Health, Dental, Disability, Life & Unemp	\$	105,161	\$	88,617	\$	16,544	84.279
7055 · Retirement Fund	\$	45,636	\$	43,035	\$	2,601	94.309
7057 · Payroll Taxes	\$	40,303	\$	34,781	\$	5,522	86.309
7060 · Printing & Postage	\$	1,050	\$	391	\$	659	37.21
7069 · Allowance for Direct Grant Expenses					\$	-	
7070 · Professional Services	\$	48,220	\$	10,536	\$	37,684	21.85
7071 · Audit	\$	7,000	\$	7,038	\$	(38)	100.549
7072 Contracted Services	\$	40,000	\$	2,892	\$	37,108	7.239
7074 · Legal	\$	500	_		\$	500	0.009
7075 · Payroll Service	\$	720	\$	606	\$	114	84.179
7080 · Rent	\$	9,672	\$	8,866	\$	806	91.679
7090 · Traffic Equipment	\$	8,000	\$	563	\$	7,437	7.04%
7095 · SADES Equipment	\$	-	\$	•	\$	-	#DIV/0!
8010 · Travel Expense	\$	4,500	\$	3,359	\$	1,141	74.64%
8050 · Utilities	\$	15,364	\$	14,373	\$	991	93.55
8051 · Electricity and Propane	\$	6,000	\$	4,211	\$	1,789	70.199
8052 · Telephone and Internet	\$	9,364	\$	10,162	\$	(798)	108.529
8060 · Vehicle O&M	\$	5,500	\$	3,606	\$	1,894	65.569
S	\$	965,201	\$	818,520	\$	151,232	84.80
	A.					Target:	91,679

Attachment 3



**Lakes Region Planning Commission** 

# **Monthly Executive Report**

# FY24 May 2024

## Local

- Andover, Gilmanton, Moultonborough & Wolfeboro HMP. Submitted invoices for all four communities. Developed agendas and worked on hazard research. Corresponded with colleague and EMDs in Wolfeboro and Andover regarding committee and plan development steps.
- **Center Harbor.** Reviewed site plan regulations and attended Center Harbor Planning Board meeting. Responded to various questions after meeting.
- Freedom. Responded to a question from the Planning Board Chair regarding LRPC services. Printed land use maps for Freedom Conservation Commission; Land Use/Land Cover map (Figure 1), Wildlife Priority map, and Natural Resources map.
- Gilford SADES. Worked on data and mapping. Met with Gilford DPW Director regarding scope, expectations, status, local issues, and final products. Field work will begin in early June. Combined Gilford's GIS data layer with LRPC's 2016 data layer for Gilford to create a comprehensive map of the locations needed for CCDS data collection (Figure 2).
- New Hampton. Provided list of planning consultants to Planning Board member for master plan update.
- Northfield. Responded to a question from the Town Administrator concerning lot merger.
- Ossipee. Reviewed letter from Ossipee concerning a possible Development of Regional Impact (DRI). None found.
- **Tamworth.** Researched information for Tamworth Sewer Commission regarding community septic/sewer systems and reviewed report on septic management prepared by LRPC for Moultonborough.
- Reviewed various municipal master plans to determine demographic data needs.





## **Grant Administration**

- Reviewed guidelines and requirements for amendment for Microenterprise Technical Assistant Program.
- Assisted Granite State Adaptive with the environmental review and RFQ for construction manager.
- Requested Amendment to budget/scope for WEDCO's Microenterprise Technical Assistant Program.

### **Household Hazardous Waste**

• Fifteen of twenty-one communities have paid invoices totaling \$66K. Conducted site visits in Tuftonoboro, Meredith, and Bristol with Site Coordinators, stakeholders, and Tradebe staff. Distributed annual HHW collection event press release #1. Held second Coordinators Meeting.

## **Economic Development**

• Regional Plan Update HUD. Worked on CEDS document updates, and compiling housing & demographics data.

## Regional

- NHDES | Source Water Protection Belmont, Northfield, and Sandwich). Communicated with DES regarding the Model Groundwater Protection Ordinance. Spoke with Belmont Planner concerning the Town's decision not to proceed with updating their Groundwater Protection Ordinance. Provided final outreach and information materials to Northfield. Submitted final report and deliverables to DES for completing grant.
- **PRLAC.** Assisted group with distribution of new Corridor Management Plan and preparations for display table at LRPC Annual Meeting.

## Solid Waste

- Participated in Tamworth transfer station site walk through to discuss possible upgrades including traffic flow review, scale design/placement, scale house design/placement, stormwater management structures for waste trailers, where to site a storage building, bathroom remodeling, and what would be needed for an eye wash station.
- Discussed capped landfill monitoring and training with NHDES. Discussed possible guest speaker for training.
- Researched scale construction and certifications. Obtained quotes for potential scale and bailer purchase.
- Designed and ordered banners and t-shirts for HHW collection event.
- Discussion with Bristol recycling center supervisor concerning cardboard bailing and associated costs.
- Completed site review plan, pay-back period, and investment options report for Tamworth recycling facility project.
- Attended in-person meeting concerning Laconia Phase 1 Brownfields sites.

## **Transportation**

- **Traffic Counting.** Counts for the season have begun and will continue most Tuesdays through September. Local police and DPWs are being notified of the work. Deployed and retrieved traffic counters in Belmont, Gilmanton, Tilton, and Northfield. Processed data and submitted to DOT.
- CMAQ. Attended May meeting. Air quality analysis for EV charging station projects by RPCs was discussed, and modelling parameters determined. Analysis is to be completed by the end of August (Plymouth).
- Complete Streets. Attended May meeting of Complete Streets Advisory Committee.

- HSIP. Attended May meeting. Rumble strip work along NH 106 will begin in June. Local projects and Road Safety Audits (RSAs) were discussed.
- Attended May interagency meeting.
- **SADES.** Created new GIS maps for CCDS to be used in the field.

## Training

- Attended training for both Stream Crossing & Culverts and Closed Drainage Systems collections.
- Attended 2-day brownfields training event in Portland, Maine.

# ACRONYMS

CCDS	Culverts & Closed Drainage Systems
CEDS	Comprehensive Economic Development Stragegy
CMAQ	Congestion Mitigation Air Quality
DPW	Department of Public Works
EMD	Emergency Management Director
EV	Electric Vehicle
GIS	Geographic Information System
HHW	Household Hazardous Waste
HMP	Hazard Mitigation Plan
HSIP	Highway Safety Improvement Program
HUD	Housing & Urban Development
NHDES	NH Department of Environmental Services
NHDOT	NH Department of Transportation
PRLAC	Pemigewasset River Local Advisory Committee
RFQ	Request for Quote
RPCs	Regional Planning Commissions
SADES	Statewide Asset Data Exchange System
WEDCO	Wentworth Economic Development Council

Attachment 4

2023 Lakes Region

# Bicycle and Pedestrian Plan

**Bicycling and Walking:** 

# Transportation Choices for New Hampshire's Lakes Region



August 2023

# www.lakesrpc.nh.gov

This report was funded in part through a grant from the Federal Highway Administration, U.S. Department of Transportation (U.S. DOT) and administered by the New Hampshire Department of Transportation (NH DOT). The views and opinions expressed herein do not necessarily state or reflect those of the U.S. DOT or the NH DOT.





**Belmont Roderick Cameron** 

TBD

Andover

Ashland

Barnstead

David Kerr

Effingham Hebron Ivan Quinchia

Moultonborough Cristina Ashjian Celeste Burns

Sanbornton TBD

Wolfeboro Roger F. Murray, III Travis Austin

**FY24 STAFF** 

Jeffrey R. Hayes **Executive Director** 

Matthew Rose Solid Waste Planner

**Ryan Paterson** Assistant Planner Carl Carder Finance Administrator

**Christine Marion** Land Use Planner

Alexandra Lermond Transportation Intern **David Jeffers Regional Planner** 

Tracey Secula Grants Administrator

Allen Constant Transportation Technician Sean Chamberlin Sr. Transportation Planner

Linda Waldron Executive Administrative Assistant











# **TABLE OF CONTENTS**

Executive Summary	V
Chapter 1: Context	1
Plan Purpose	1
Geographic and Demographic Context	2
Planning History	3
Why a Bicycle and Pedestrian Plan for the Lakes Region?	4
Chapter 2: A Vision for a Bikeable & Walkable Region	5
Revising the Vision	5
Clarification of the Vision Statement	6
Goals of this Plan	6
Chapter 3: Plan Development and Community Input	9
Public Survey	9
Design Concepts	9
Chapter 4: Existing Conditions	11
Multi-Use Trail Development	11
Parking	14
Pavement Conditions	16
Shoulder Space	17
Lane Widths	19
Bikeability Assessment	19
Bicycle Routes and Maps	23
Chapter 5: Use Trends and Public Input	24
NH and US Data and Policies	24
Concerns Expressed Through Public Meetings and Survey	24
Bicycle and Pedestrian Data Collection	27
Walking Data	28
Bicycling Data	29
Chapter 6: Bike and Pedestrian Projects	30
Overview	30
Recently Completed Projects	30
Upcoming Projects	33
Project Recommendations	35









Chapter 7: Action Plan to Support the Vision	
Strategies and Actions	41
Appendices	
Appendix A: Historical Bike and Pedestrian Documents in the Lakes Region	43
Appendix B: Survey Questions and Results	49
Appendix C: Summarized Local Master Plans and Goals	55
Appendix D: Maps	62





# **EXECUTIVE SUMMARY**

The Lakes Region is home to more than 125,000 residents and is a thriving area to visit during the beautiful summer months where many travel to stay on the beautiful lakes New Hampshire has to offer. The Lakes Region surrounds the beautiful Lake Winnipesaukee and many smaller lakes, ponds, and streams. The 31 communities sit at the center of New Hampshire separating the Northern half of New Hampshire and the Southern half. The Region is made up of small rural communities, the largest city in the Lakes Region has a population of less than 20,000.

The Lakes Region municipalities are striving towards creating infrastructure that allows those to continue exploring the beautiful landscape by walking and bicycling. For decades, motorized vehicles have been the dominant form of travel in the Lakes Region. There are only small amounts of sidewalks and designated bike lanes throughout the region that allow those to travel through the main downtown/village centers of the towns and cities. The 2023 Lakes Region Bicycle and Pedestrian Plan is a comprehensive update to the 2012 plan. It was developed to provide the communities in the Lakes Region with additional information about existing biking and walking infrastructure and to reinforce local revitalization efforts that desire to make their communities safer and more enjoyable places to live and visit.

The planning process for this report included public participation for a new vision statement, a review of existing conditions of the built infrastructure, and recommendations.

### The 2023 Vision Statement:

A purposefully connected network of trails, sidewalks, road shoulders, and markings promoting safe, enjoyable, and inclusive bicycle and pedestrian mobility for residents of all ages and physical ability. Design and maintenance of livable, complete streets that support transportation, recreation, health, and economic interests throughout the Lakes Region.

Implementation of transportation improvements such as road reconstruction, bike paths, sidewalks, etc. often takes many years, from a project's purpose and need to final construction. The long view is important. The region is in the process of developing an updated transportation plan and process for motorized travel and currently has a well-developed plan available. Frequently, roadway funding and facility decisions are made with little consideration of the needs of bicycling and pedestrians. This plan is an attempt to recognize bicycling and walking as serious transportation modes, and to encourage our communities, and the state, to integrate them into future improvements of the transportation network. With recent technological developments over the years and growing infrastructure, bicycling and walking are becoming a more popular mode of travel compared to the traditional motorized mode.



Chapter 1 outlines some of the important reasons to support walking and bicycling opportunities in the Lakes Region.

- The population of the Lakes Region grew by 5.12% from 2010 to 2020. This means trails, bike lanes, and all corridors of pedestrian transportation will continue to see increased use but need continued expansion and updates to keep up with the growing population.
- The median age of the Lakes Region is estimated to be 49.8 – eleven years older than the US median age and seven years older than New Hampshire's median age. Active transportation can support health and mobility, allowing people to age in place and participate in the local economy for as long as possible.
- Gasoline prices are continuing to rise in the aftermath of the COVID-19 pandemic as demand for fuel soars. Motorized transportation has become increasingly expensive in a time of returning to normalcy after nearly two years of pandemic stagnation.

Chapter 2 describes the development of the vision statement and the goals supporting the vision and describes the changes from the 2012 plan to the 2023 plan.

Chapter 3 presents public input obtained at three public meetings that were held throughout the Lakes Region near the end of 2020 and the beginning of 2021. The meetings were facilitated to allow people to express needs and concerns related to bicycling and walking, and to present their ideas for solutions. Input was summarized into topical statements of need/concern. In addition, a public survey was available from December 2021-January 2022 for the public to answer a variety of bicycle and pedestrian questions.

Chapter 4 examines the existing conditions of the built environment in the Lakes Region as they affect pedestrian and bicycle safety and accessibility and includes discussion of the potential for improvements to be made. Parking, intersections, crosswalk, trails, sidewalks, travel lanes, shoulder width, pavement conditions, and maps are all discussed. Research on pedestrian and bicycle safety and road design is also presented.

Chapter 5 then goes into an analysis of the data mentioned in Chapter 3. Past policies relating to bike and pedestrian usage are outlined along with relevant crash data. Chapter 5 also includes a breakdown of the survey results in order to determine recent trends in bike and pedestrian activities. This will help give a sense of what people are looking for in terms of new infrastructure and policies.

Chapter 6 gives examples of regional bike and pedestrian projects at different stages. Recently completed projects provide a sense of what is being done successfully and what can be improved upon in future projects. The list of upcoming projects then creates a snapshot of what is to be expected in bike and pedestrian infrastructure in the near future. The final section includes project recommendations for the Lakes Region. These are based off recent proposals and how well the project aligns with the goals of this report. Lakes Region Planning Commission (LRPC) provides guidance on the next steps of bike and pedestrian improvements.

Chapter 7 outlines an action plan for the region that will help bring the 2023 vision to life. Hopefully, the strategies listed can hold the LRPC and its communities accountable in prioritizing bike and pedestrian infrastructure.



# **CHAPTER 1: CONTEXT**

Walking, or assisted mobility, is the most basic form of transportation. All modes of transport, whether it be taking public transit, driving a motor vehicle, or bicycling require walking at some point. Despite this, walking is often overlooked as an "alternative" mode of transportation and not considered as much as motor vehicles in transportation planning. All trips require a pedestrian element (walking at destinations and walking from origins to other modes of transportation) Walking, or assisted mobility, is a fundamental aspect of daily life and society that is typically overlooked.

The reason walking, or assisted mobility, is not easily addressed under any one topic is that it is an inherent element of economic activity, recreation, physical and mental health, social interaction, and vibrant community. Consider a business who is resistant to roadway improvements that reduce street parking, only to notice it increases the number of visitors as a result of a more inviting and walkable public place; it is the people who spend the money, not the number of vehicles that travel past the business. People are what our communities are built around and providing ways to bring ourselves and our services together, to work, to school, to the lake, and any place is how our communities are shaped.

To accommodate bicycle and pedestrian needs, we must acknowledge the fundamental roles that pedestrians and bicycles play in the equation. We must take a look at what is needed for bicyclists and pedestrians. What infrastructure do they need to get from Point A to Point B. What will best protect pedestrians and bicyclists from motorized travelers. How will the bicyclists and pedestrians know where infrastructure is available to use? Safe opportunities to move about our communities while not in cars must be provided for and maintained in our communities. This plan outlines the case of continuation of many good efforts to enhance pedestrian and bicycle resources that have been underway around the region and present the cases for more deliberate consideration of our public streets as assets for supporting economic, physical, and social health of our communities.

# **Plan Purpose**

The intent of the 2023 Lakes Region Bicycle and Pedestrian Plan is to build on what is presented in the 2012 Bicycling and Walking: Transportation Choices for NH's Lakes Region, and the achievements of the communities and organizations over the past 11 years. This plan will expand on the recommendations for achieving the 2023 Vision and highlight what changes the Lakes Region has seen over the past decade on expanding and improving bicycling and walking becoming a more prominent mode of transportation and activity in the growing Lakes Region.

The purpose of this plan is to provide a framework to guide policies of the LRPC related to bicycle and pedestrian planning. This framework will help LRPC and the communities in the lake's region develop comprehensive transportation plans that include non-motorized transportation planning infrastructure developments and help communities see what opportunities are available to grow on bicycling and walking. The intent is to raise awareness among local and state officials of the stated needs and priority interests of residents and visitors to the region related to bicycle and pedestrian safety and accessibility. This plan



is designed to be used communities as a resource as they develop their own local bicycle and pedestrian plans and to be used as a resource when they develop their own projects designed for non-motorized travel.

# Geographic and Demographic Context

The thirty-one (31) municipalities of the Lakes Region comprise a diverse area of land use and activities including many yearround outdoor recreational opportunities. Most of the lakes and ponds are surrounded by a mixture of resorts, summer cottages, second homes, and year-round homes. Many of the roads that encircle them are close to the shore. This mix of residential developments transforms the smaller rural towns into busy and often congested communities during the peak summer tourist months. Many of the Lakes Region corridors see double the amount of vehicular traffic during the busy summer months from those who travel to the beautiful lakes and ponds in the region. In addition to the influx of motor vehicles, onroad cyclists, from individuals, families, and children, can be seen pedaling around the beautiful lakes in the region. The roads see an increase in use from those biking to villages, recreational areas, downtown destinations, and more.

Squam Lake and Lake Winnipesaukee split the Lakes Region into two parts, with eight (8) towns laying to the north and east of these lakes and the other twenty-three (23) towns to the west and south. The Region also borders the White Mountain National Forest to the north, including the Squam Mountains and the Sandwich Range. The Ossipee Mountains to the northeast of Lake Winnipesaukee, and the Belknap Mountains to the south of Lake Winnipesaukee create additional challenges for transportation through these areas. In the Lakes Region, as is the case throughout much of New Hampshire, north-south travel tends to be much easier than east-west travel, primarily because of natural barriers created by lakes and the orientation of the mountains. Consequently, there are many roads in the region where no convenient alternative exists for bicyclists and pedestrians to travel along safer, lower vehicle-volume routes.

### The Lakes Region is Aging

Similar to much of New Hampshire, the Lakes Region communities are aging. The median age of year-round residents of the Lakes Region in 2021 was 49.8. This is nine years (33.8 percent) older than the national median age of 37.2. It is also more than seven years older than the state median age. The regional median age by town now ranges from over 57 in Hebron to just under 24 in Plymouth.



In addition to year-round residents, the Lakes Region is home to many thousands of summer seasonal households, the majority of which are either families with young children or more elderly, retired people. Lakes Region communities The are attracting more and more retirees from out of state, seeking a place that can provide outdoor and active recreational resources. The number of older residents is significant to consider in this regional bicycle and walking plan because having safe and accessible walking and bicycling



opportunities can support independent mobility longer than motor vehicle dependence for some.

# **Planning History**

The New Hampshire Department of Transportation (NHDOT) developed the first pedestrian and bicycle plan for the state in the 1970s. This plan was updated first in 1995, again later in 2000, and most recently in the fall of 2023. The New Hampshire Statewide Pedestrian and Bicycle Plan was developed as an element of the comprehensive New Hampshire Long Range Statewide Transportation Plan. Its purpose is to meet the requirements and needs for promoting all modes of transportation as established by the Intermodal Surface Transportation Efficiency Act (ISTEA) of 1991 and reaffirmed first by the Transportation Equity Act for the 21st Century (TEA-21) of 1998 and later by the Safe, Accountable, Flexible, Efficient Transportation Equity Act: A Legacy for Users (SAFETEA-LU) of 2005.

The vision statement outlined in the 2023 New Hampshire Statewide Pedestrian and Bicycle Plan states that "The State of New Hampshire will enhance safety and mobility through improved pedestrian and bicyclerelated policies that continually expand and enhance our network of pedestrian, bicycle, and transportation facilities designed for a wide variety of users and abilities." Although the Plan is an information resource for regional and local planning, its primary focus is on meeting the needs for bicycle and pedestrian planning at the state level. For example, the Plan discusses the need to connect major destinations throughout the state and focuses on the ability of the state transportation system to do this. As such, the focus of the state plan is on improvements and strategies related to the

numbered and unnumbered state routes throughout New Hampshire.

In spring 2001 the NHDOT released a new map of the "Statewide Bicycle Route System". This map was developed through the New Hampshire Bicycle and Pedestrian Transportation Advisory Committee, an appointed group of public officials and organized bicyclists. Input from many experienced and locally knowledgeable cvclists was received during the informational meeting process and incorporated into the route map. The purpose of this map was to identify the best highway routes for bicyclists wishing to travel from one major destination to another. However, these maps do not necessarily reflect the safest routes available. Although safety was a factor, logical transportation routes and distances were the primary considerations in determining these routes. The State and Regional Bicycle Routes map provides a valuable tool from which to base bicycle planning in the Lakes Region.

The comprehensive transportation policy plan for the Lakes Region, PLAN 2000, was adopted by the Lakes Region Planning Commission in September 2000. The overarching principal of PLAN 2000 is "the provision of efficient multi-modal and intermodal transportation systems which do not negatively impact the surrounding natural, social or historic environments." Due to its comprehensive nature. PLAN 2000 addressed only improvement needs for the region at a very general level, identifying possible bike routes throughout the Region.

The 2012 plan highlighted the successful ways that communities in the Lakes Region have accommodated and promoted pedestrian and bicycle travel. It also articulated a vision for the successful completion of an inter-connected off-road multi-use pathway spanning the region, outlined a process for local pedestrian and



bicycle planning, and discussed some of the important land use planning tools that can support implementation. While the vision and goals of the 2012 Plan are largely still relevant and appropriate, this comprehensive update includes an assessment of bicycle and pedestrian safety, accessibility, and facilities. It also includes an expanded public engagement process targeted at understanding the needs and concerns of residents of the region, and the development of a more comprehensive set of recommendations for the LRPC. communities, and others to consider.

In 2015, LRPC adopted an updated regional transportation plan. lt includes comprehensive set of recommendations but did not address non-motorized needs in great detail. 2023 Lakes Region Bicycle and Pedestrian Plan is an important component of the regional transportation program and serves to augment the 2015 Transportation through consideration of Plan nonmotorized modes. See Appendix A for more detailed information on relevant state and regional planning efforts.



# Why a Bicycle and Pedestrian Plan for the Lakes Region?

An overarching reason to plan for walking and bicycling is because many people want to feel safer doing so in their communities. Families and individuals, young and old, recreational and utilitarian users across the region have expressed not only a sense of concern for personal safety and the safety of children, but have also expressed a strong desire for more walkable and livable communities. When people feel safe, and perceive walking and bicycling as enjoyable activities, they spend more time in public places, whether shopping, strolling, bicycling for fun or exercise, being with others, or simply enjoying the outdoors. Safe and well-designed public places leverage desirable community activity, and they provide opportunities to off-set some other health and environmental impacts created in part by our dependence on motor-vehicles.



# CHAPTER 2: A VISION FOR A BIKEABLE & WALKABLE REGION

"Livability means being able to take your kids to school, go to work, see a doctor, drop by the grocery store or post office, go out to dinner and a movie, and play with your kids in the park – all without having to get in your car."

~U.S. Transportation Secretary, Ray LaHood

# **Revising the Vision**

The vision statement of the 2012 Plan was revised based on review and discussion from Lakes Region Staff and the Transportation Advisory Committee in the Spring of 2022, and the input from the Lakes Region Residents through the Bicycle and Pedestrian Plan.

### The 2012 Vision:

"A purposefully connected network of trails, sidewalks, road shoulders, and markings promoting safe and enjoyable bicycle and pedestrian mobility. Design and maintenance of livable, complete streets that support transportation, recreation, health, and economic interests throughout the Lakes Region."

Meeting participants generally agreed with the vision statement from 2012 but wanted it to convey the importance of personal comfort and enjoyment derived from being able to safely and easily walk and bike. While biking and walking can always be viewed as travel and transportation, these terms may suggest a purely utilitarian idea of biking and walking. The term 'mobility' is considered to be a more inclusive term that does not suggest any one purpose.

Additionally, participants agreed that their vision was of an environment where biking and walking were actively promoted or deliberately supported through design and function, and that the many elements of a non-motorized pathway network be explicitly included as part of the vision. This was mainly a reflection of participants' perception that the former plan focused primarily on improving the multi-use trails of the region. Lastly, the majority of participants found the use of the word 'feasible' to be confusing, assuming that it implied that the vision was for bicycle and pedestrian accommodation only where feasible. While the term 'feasible' was used in 2006 to indicate a vision for an environment where biking and walking is a feasible option, most participants who critiqued the 2006 vision felt that a safe and enjoyable network necessarily included feasible options to walk and bike.

### The 2023 Vision:

"A purposefully connected network of trails, sidewalks, road shoulders, and markings promoting safe, enjoyable, and inclusive bicycle and pedestrian mobility for residents of all ages and physical ability. Design and maintenance of livable, complete streets that support transportation, recreation,



# health, and economic interests throughout the Lakes Region."

This plan is designed to achieve the vision by promoting safe bicycle and pedestrian mobility throughout the region while supporting the construction of inter- and intra-town pathway systems by providing the public and non-public sectors with guidance and data for funding, design concepts, and technical contacts.

# Clarification of the Vision Statement

### Safe

Bicycle and pedestrian travel become safe when motor vehicle hazards to nonmotorized users have been minimized, when problem areas along the network have been addressed through the provision of adequate space or corrections to design, when built sections have been connected, non-motorized and motorized users have been educated, and enforcement is an ongoing process.

### Enjoyable

Bicycle and pedestrian travel become enjoyable when it is safe, and when it is apparent to the user that pedestrian and bicycle needs have been thoughtfully considered in the design, construction, and maintenance of the pathway network, particularly with regard to exposure to motor vehicle hazards, and when useful signage and informational materials have been provided that identify routes, hazards, and helpful contextual information.

### Inclusive

This term is added to the 2023 vision statement in order to encourage design and consideration of projects that improve accessibility for all residents in the Lakes Region. As it is further outlined in the vision statement, this includes residents of any age, gender, and physical ability. The Lakes Region will strive to create a cohesive bike and pedestrian infrastructure focused around the principles of equity and inclusion.

The updated vision is supported by the following goals and objectives.

**GOAL 1 – CONNECTIVITY**: Improve connections between the region's walking and bicycling routes.

- Complete regional and local connections through on-street networks that more seamlessly link city and town centers with rural areas.
- Eliminate gaps in the rail trails and shared use path networks.
- Promote better links between destinations with improved pedestrian and bicycle facilities.
- Support and encourage efforts of the NHDOT and local public works departments to increase bicycle infrastructure.



**GOAL 2** – **DESIGN:** Collaborate with NHDOT and the region's municipalities on bicycling and pedestrian infrastructure design that reduces level of stress experienced by bicyclists and walkers.

- Plan for sidewalks and safe crossings where warranted on state highways and local roads that that run through areas with concentrated commercial and/or residential land uses.
- Encourage bicycle facilities designed for "all ages and abilities" that incorporate traffic calming elements (e.g. narrower lanes, on-street parking with bump-outs, roundabouts, raised speed reduction medians, horizontal and vertical deflection, changing one-way streets to two-way, etc.).
- Participate in the NHDOT's Complete Street Advisory Committee.
- Use the Lakes Region Transportation Technical Advisory Committee (TAC) as an opportunity to communicate with local communities about transportation bicycling and walking.
- Establish a Lakes Region Bicycle and Pedestrian Advisory Committee as a standing committee to the Transportation Advisory Committee to monitor plan recommendations on an annual basis, and to update the plan at least every 4-5 years.

**GOAL 3** – **POLICY:** Provide outreach and training to interested Lakes Region municipalities on streetscaping and complete streets policies that promote walking and bicycling.

- Collaborate with Lakes Region municipalities on Transportation Alternatives Program (TAP) Project proposals.
- Distribute bicycle education and encouragement programs to interested Lakes Region communities.
- Provide Lakes Region communities with information and materials on land use strategies and regulations which address bicycle safety and usage.
- Provide communities and interested organizations with useful regional walking and bicycling maps and information.

GOAL 4 – EQUITY: Promote equity in pedestrian and bicycle planning.

- Ensure that bicycle and pedestrian planning includes understanding the needs in traditionally underserved communities.
- Encourage funding that takes into account need, safety hot spots and future demand.
- Promote the design of bicycle and pedestrian infrastructure that improves accessibility for users of any age and physical ability.



**Goal 5 - DATA:** Ensure a continuous and comprehensive process of bicycle and pedestrian data collection.

- Communicate with, and assist interested Lakes Region communities with collecting accident, geographic, trip and other related data.
- Maintain research files on bicycle and pedestrian related information.
- Coordinate data collection with NHDOT, other governmental and non-governmental organizations.
- Encourage municipalities to conduct before and after studies of new bicycle facilities to measure effectiveness.

**Goal 6 - SAFETY:** Ensure that there are opportunities for the safe movement of non-motorized modes of travel.

- Encourage a continuous, comprehensive program which first addresses the most viable on-road and off-road routes.
- Support local programs, education, and local law enforcement efforts that are designed to reduce the number of bicycle and pedestrian accidents, injuries, and fatalities, particularly those that involve motorists.
- Support a policy of creating usable shoulder widths for bicycles and pedestrians on rural roads.
- Prioritize pedestrian and bicycle access and safety along roads near public or private K-12 school.
- Encourage flexible speed-limit policies that increase safety on state highways.

**GOAL 7** – **ECONOMY and HEALTH:** Leverage walking and bicycling improvements to promote economic development and improve public health opportunities.

- Encourage local land-use policies and design strategies that encourage walkability and bicycling.
- Partner with public health organizations and officials to identify data needs to measure the impacts of walking and bicycling on public health.
- Facilitating bicycling and pedestrian improvements along the Lakes Region Tour Scenic Byway.
- Support local and statewide educational and incentive programs that encourage people of all ages to walk and to ride bicycles for health and mobility.
- Promote visibility and wayfinding of pedestrian and bicycle infrastructure.
- Increase cooperation between municipalities in creating and marking pedestrian and bicycle infrastructure.



# CHAPTER 3: PLAN DEVELOPMENT AND COMMUNITY INPUT

The development of the *2023 Lakes Region Bicycle and Pedestrian Plan* began in the winter of 2021.

An update of the 2012 plan was viewed as an important step to enhancing the existing bicycle and pedestrian network and promoting the Lakes Region as a biking and walking destination. LRPC's vision is to have a plan that is used by community leaders, public officials, regional tourism agencies and biking and walking enthusiasts in order to promote walking and biking as viable and healthy mobility options, and to promote the development and preservation of suitable infrastructure for safety and enjoyment.

## **Public Survey**

An internet-based survey was designed to collect data on biking and walking, and to provide an opportunity for those who could not attend the public meetings to be able to provide input on biking and walking in the region. The survey was publicized similarly to the meetings.

The survey was open for six weeks from the beginning of January to Mid-February. In that time frame, 634 responses were received. Survey questions were designed to understand:

- For what purposes people bike and/or walk
- What facilities people use to bike and/or walk

- What prevents people from biking and/or walking more
- What would encourage people to bike and/or walk more

Specific survey results are discussed in detail in Chapter 5, where they serve to highlight existing conditions and views about walking and bicycling. The complete survey results can be found in Appendix B.

## **Design Concepts**

In support of a more comprehensive assessment of the existing conditions for bicycling and walking, LRPC solicited nominations from member municipalities for sites throughout the region that were particularly unsafe or poorly accommodate pedestrians and bicycles. The intent was to highlight a number of sites that are representative of different kinds of design problems, and that may share commonalties with many other sites in terms of existing barriers to safe and enjoyable accessibility as well as potential solutions.

Respondents were asked to nominate intersections, road segments, network connectivity gaps, or trail related sites that were either unsafe or presented access challenges. LRPC asked for a descriptive narrative and for supportive documentation that would help describe and characterize the sites. LRPC emphasized that sites that were of regional significance would be prioritized for review.



As part of the effort to characterize existing conditions in the Lakes Region, LRPC engaged Resource Systems Group, Inc., a transportation planning firm with offices in Concord, NH, and Ironwood Design Group, LLC, a landscape architecture, urban design, and environmental planning firm located in Exeter, NH, to assess bicycle and pedestrian access and safety improvements in 15 municipalities in the Lakes Region and provide graphics depicting conceptual solutions at 15 of 36 sites that were nominated. Site selections were made by LRPC transportation staff and were based on the following criteria:

- Regional importance
- Geographic distribution
- Severity of existing conditions
- Nature of sites (trail, sidewalk, intersection, right-of way constraint, etc.)

Consideration was given to a balanced selection of rural and urban sites when developing a priority list of sites. LRPC staff and consultants teamed with municipal officials in conducting field assessments at each location. The team of engineering and design professionals assessed each of the sites that were selected. Sites were visited over three days with LRPC staff and local representatives familiar with the particular challenges to safe and enjoyable biking and walking. Following site visits, the team met to review findings and discuss conceptual solutions that would address the challenges that were observed and conveyed by local stakeholders.

Detailed site analysis and design improvement recommendations are presented in a special supplement to this report titled, Bicycling & Walking: Conceptual Design Supplement. It contains detailed narrative assessment of existing conditions at each of the 15 sites, and discussion of conceptual design improvements that address the accessibility and safety problems at each site. Existing conditions are depicted with photographs, and conceptual designs are illustrated to show how pedestrian and bicycle safety and access could be improved. The challenges present at these sites are related to many of the topics discussed in Chapter 4 (Review of Existing Conditions) and help to highlight the conditions that must be addressed in order to be more walkable and bikeable, and livable.



# **CHAPTER 4: EXISTING CONDITIONS**

There are many elements of the built environment that affect pedestrians and bicycles, and the policies that determine the design and function of these elements are just as important. This chapter reviews the condition and impacts of many important elements of the built environment in the Lakes Region. Design and policy consideration related to these elements of the built environment are discussed as well.

# **Multi-Use Trail Development**

Trail conditions in the region are viewed as good by survey respondents. They generally offer more space than sidewalks, a significantly reduced exposure to motor vehicle hazards. and more intimate access to natural surroundings. More trails completed and connections were prioritized by both survey respondents and public meeting attendees, and are considered to be a priority regional goal. LRPC should continue to work with NHDOT and Hampshire New Department of Natural and Cultural Resources (DNCR) to support policies that will help build successful trails. LRPC should also continue trail work with to organizations to support economic development studies, planning and design considerations, and federal funding for

project completion. Since 2006, sections of multi-use trail have been completed throughout the Lakes Region.

### WOW Trail – 4.5 miles

The first phase of the Winnipesaukee Opechee Winnisquam (WOW) Trail was completed in early 2010. The paved path connects bikers, walkers, and runners from Elm Street in Lakeport to downtown Laconia. It became a popular trail through



Map of proposed Opechee Loop trail



the heart of the Lakes Region's most densely populated community. In fact, the trail was the most used one among survey respondents. Phase two was completed in the fall of 2016. It continues the trail from Laconia's train station to a convergence with Belmont's Winnisquam Scenic Trail at Belmont's Leslie Robert's Beach before ending at Lakeport Square at the corner of Elm St. and Union Ave in Laconia. The trail offers beautiful views of Lake Winnisguam while providing opportunities for swimming and picnicking along the route. In total, phases I and II of the WOW Trail and Belmont's Winnisquam Scenic Trail create 4.5 miles of continuous trail for pedestrians and cyclists to enjoy.

With phases I and II of the trail being so popular, in October 2021 plans were revealed to extend the WOW Trail by five miles around Opechee Bay. The extension would begin at Lakeport Square and travel perpendicular to Elm St. through the former State School property. Then, it would loop around to travel through Ahern State Park on the other side of the bay, before continuing to run perpendicular to the entire length of Shore Drive. At that point, it would travel along Holman St, Gale Avenue, and finally a three-block section of Pleasant St. before reconnecting to the lower portion of the Rail Trail in Veterans Square. This extension would add an additional 2.5 miles of trail, totaling the WOW Trail to seven miles of non-motorized recreation for the community to enjoy.

Over the past decade, there have also been plans to extend the trail from Lakeport north to Weirs Beach. However, this project has been difficult to complete due to opposition from abutters from the South Down and Long Bay condominium communities.

### Winnisquam Scenic Trail

The Winnisquam Scenic Trail will eventually be 5.2 miles long and will connect the WOW Trail in Laconia and the Winnipesaukee River Trail in Tilton. Phase I of III of the trail was completed in Belmont in August 2016. It begins at Leslie E. Roberts Drive on the Laconia-Belmont city line, turns on to Horse Point Road, then continues through the woods on NHDOT land. The paved path runs along the former B&M Railroad along the shore of Lake Winnisquam, crossing a trail easement donated by Sun Lake Village. After 1.8 miles, phase I ends at US route 3 above Mosquito Bridge.

Phase II of the Winnisquam Scenic Trail is currently in the planning stages. If completed, it would connect an extra 0.5 miles to the existing 4.5 miles of continuous paved trail between the Lake Winnisguam Scenic Trail in Belmont and the WOW Trail in Laconia. Phase II would begin at Osborne' Agway Garden Center, the end of the existing trail segment before reaching US Route 3. The proposal includes a pedestrian and bicycle hybrid traffic light to help pedestrians manage the busy road. The exact location for the crossing station is ongoing with the NHDOT Traffic Bureau. From there, phase II of the trail turns parallel to Route 3 before turning south on a proposed 1,200 ft raised boardwalk adjacent to the existing railroad track between Timothy Drive and Dupont Road This boardwalk would end at Elaine Drive. The \$1.25 million project cost would be covered by local funds as well as funds from Federal Highway the Transportation Alternative Program.

The challenge in completing this addition is the need to cross US Route 3. The current speed limit would create an unsafe environment for pedestrians and cyclists, and it would likely have to be reduced.


#### Winnipesaukee River Trail – 5.1 miles

The Winnipesaukee River Trail provides trail access from the city of Franklin and the Northern Rail Trail, eastward to Tilton, Northfield, and the ongoing trail projects in Belmont and Laconia. It is a completed section of the 2006 vision of the Lakes Region Connector, which would allow for trail travel throughout the Lakes Region. In 2011, steps were taken to add a connection across the Winnipesaukee River between Tilton and Northfield, but as of 2021 there are no longer plans to install one.

#### Northern Rail Trail - 58 miles

The Northern Rail Trail is part of the longest rail trail in New Hampshire spanning 58 miles from Lebanon to Boscawen. It offers scenery of wetlands and wooded areas, and views of the Merrimack and Mascoma Rivers. The elevation grade is mostly never greater than 1% and the surface consists of cinder or gravel.

26 miles of this trail passes through the Lakes Region communities of Danbury, Andover, and Franklin. As of 2021, the Merrimack County portion of the trail is 34 miles long. Over the past decade, a total of 9.3 miles of trail have been added, extending the trail south within a half-mile of the Hannah Duston Park and Ride on US Route 4 in Boscawen. In addition, volunteers have repaired culverts, bridge decking and railings, and mileposts to maintain and beautify the trail. Trail signage and a restoration of the historic round table site in Franklin was also completed.

The priority is to finish connecting the southern terminus, located in a Boscawen corn field, a half-mile to reach the Concord city line within the Hannah Duston Park and Ride. That extension would then connect the trail to the other trails emanating from Concord. Another hopeful extension is to connect the section of trail in west Franklin to the Winnipesaukee River Trail. The trail association will continue to promote the rail trail to garner support from businesses and residents within the trail's path. The NH Trails Bureau will also continue to work with the Northern Rail Trail to assure proper maintenance of the trail.

Another initiative by the Northern Rail Trail is the recent introduction of e-bike charging stations along the route. In 2020, local businesses in Danbury, Andover, and Boscawen offered access to their chargers for trail users. This agreement allows the trail to be used by all forms of recreational transportation and makes it more accessible to people of all physical abilities. For example, the Friends of the Northern Rail Trail are now promoting their own Cycling Without Age project. This program will allow seniors to participate in rail trail use by using an electric Trishaw with the assistance of volunteers. With the charging e-bikes, infrastructure to support the Northern Rail Trail is now able to promote accessibility and inclusion while strengthening its local sense of community.

#### Sewall Woods Trails – 3 miles

The Sewall Woods Conservation Area is a 179-acre woodland in Wolfeboro, NH preserved bv the Lakes Region Conservation Trust. The area provides multiuse trails for all experience levels and a scenic forest and wildlife habitat. The popular trail network within the woods has been around for about a decade and is a great example of a local trail project completed with privately raised funds. The area is a stand-alone recreational resource rather than a corridor with multiple access points; however, it is within a short walk of downtown Wolfeboro, making it an ideal resource that meets the needs of the Wolfeboro community better than corridor trail development. Corridor trail development must consider infrastructure right-of-way, adjacent property owners' interests, and conflicts with motor vehicle



needs. Whereas stand-alone recreational areas near downtown amenities provide



Map of Sewall Woods Trails

alternative places to walk than the very narrow shoulders along rural roads.

## **Cotton Valley Trail** - 12 miles (Sanbornville to Wolfeboro)

The Cotton Valley Rail Trail is a 12-mile trail in total; within the Lakes Region, it connects downtown Wolfeboro with parks and beaches. Its crushed stone path points east to Wakefield. The trail is heavily used yearround for walking, biking, running, wildlife viewing, fishing, cross country skiing, and snowmobiling. It is a great example of how crushed stone trail beds can successfully accommodate multiple users.

#### **Moultonborough Neck Trail**

Improvements are needed on the Moultonborough Neck Trail due to its unfortunate placement at a lower grade than an adjacent roadway. The elevation difference has led to drainage problems where debris is washed onto the paved trail. pedestrians and bicyclists would benefit from an improved and upgraded trail. Proper maintenance of this trail is especially important to Moultonborough residents because there are no nearby trails in surrounding communities.

#### **Ossipee Lake Trail**

The Ossipee Lake Trail is a route around Ossipee Lake that is ideal for bicyclists and pedestrians due to its on-road wide shoulders. However. face challenges users at narrow bridges and along segments with many side streets. To accommodate this, NHDOT has integrated two culvert replacements along the route in direct support of the future vision for additional bike lanes.

## Parking

Because free parking is viewed as an important economic development tool, it is free in most communities in the Lakes Region. This situation can exacerbate traffic congestion significantly. In some dense urban squares in big cities, cruising can amount to thirty percent (30%) of the traffic or more. Most of the year, it is not difficult to park in any Lakes Region downtown district or village. However, the Lakes Region's seasonal population expansion and high summer and weekend visitation rates can place severe constraints on parking availability in some of the densest commercial locations, such as Meredith, Wolfeboro, Laconia, and Tilton.

Communities can address parking demand and the hazards to pedestrians and bicyclists in several ways:



- Assess peak-demand parking supply. If fewer than fifteen percent (15%) of on street and municipal lot stalls are not regularly available, then additional supply or reduced demand is necessary. The first step is to assess the parking problem and understand opportunities and constraints for increasing supply.
- Assess whether local codes are effective and consider alternatives. Examples include: (a) shorter ontime regulations, street which promote off-street parking; (b) for charging on-street parking during peak demand (summer months, weekends, etc.), and many others. Many parking regulations make Sunday and evenings free, but if a demand problem exists during those times, it may be appropriate to charge( e.g., Weirs Beach). Fees not only help with turnover and reduce cruising congestion, but they also provide a marginal incentive for people to walk or bike if they perceive a monetary cost at their destination. This alone can help promote more walking and biking, and a safer streetscape.
- Conducting a comprehensive parking study to identify parking needs and safety.

Parking design also affects pedestrian and bicycle safety. Some communities have pullin angled parking. This parking configuration creates a convenience for drivers and an extreme hazard for pedestrians and bicycles - cars can pull into spaces directly from the travel lane without indication (posing a hazard to cyclists), and by design, back out into traffic. A safer configuration for angled parking is to have reverse-angled parking, as shown on the right side of the photo below. This slows

driver speeds when backing into spaces and gives them direct line-of-sight to oncoming cyclists when pulling out. Reverse-angled parking has had mixed results in recent years. Bristol tried to implement this change in the downtown area but was met with pushback from the town and ended up repainting to remove them. Laconia has had



success implementing reverse-angled parking stalls along the east and west portions of Beacon St.

In many lakefront communities and in the larger communities of the Lakes Region where parking stalls have been stripped on the road, stalls are striped right up to intersection corners, crosswalks, and curbcuts. This practice at intersections poses a significant hazard to pedestrians for several reasons and is in violation of state law requiring 20 feet of clearance from crosswalks at intersections. The NH Supplemental Design Criteria supplementing the FHWA Roundabouts: An Informational Guide (2000) for roundabout design on New Hampshire state maintained roadways makes clear that for the purposes of the law, roundabouts are considered to be intersections, and no parking is to be allowed within 20 feet of roundabout crosswalks.

Parked cars adjacent to crosswalks obscure pedestrian visibility to drivers and place pedestrians near turning and parking vehicles. By locating parking stalls in



violation of state law, municipalities may unintentionally be exposing themselves to personal injury claims from pedestrians, as well as drivers who were parked in illegal locations and whose vehicles contributed to line of sight related to accidents.

Crosswalks and intersections benefit greatly from intersection corner curb extensions, called "bulb-outs", that protect pedestrians from cars pulling out of parking spaces, make pedestrians visible to oncoming traffic, allow pedestrians to see on-coming more easily, reduce crossing traffic distances, and calm traffic approaching intersections. Bulb-outs at intersections and mid-block crossings also allow more parking stalls to be accommodated legally. Bulb-outs create additional challenges for snow removal, but this should be a secondary consideration where pedestrian safety is an issue.

Municipalities should mark parking stalls in accordance with state law or should install bulb-outs to enhance pedestrian visibility. Bulb-outs also provide an extra level of safety by placing a curb between parked cars and pedestrian crossing points. Municipalities experiencing peak demand parking supply problems may also wish to conduct parking demand analyses in order to evaluate options for ensuring adequate parking supply that do not involve creating serious hazards to pedestrians.

## **Pavement Conditions**

Pavement conditions affect bicyclists more than pedestrians. Many cyclists who attended public meetings expressed frustration with the state bicycle map for the Lakes Region depicting a bicycle route system along roads in such poor condition. Deteriorating roads are often heaved, have potholes that are very hazardous to cyclists, have longitudinal cracking that is equally hazardous, and have inconsistent road edges that can affect the usable shoulder width. All of these conditions also create a source for loose gravel that ends up collecting on the edges of the road where cyclists ride.

In 2019, NHDOT created a map of road conditions using data collected from 2018 -2019. In the third maintenance district, where most Lakes Region communities are located, the pavement conditions are classified as 45% good, 34% fair, 19% poor, and 2% very poor. With less than half of roads considered good, this limits cyclists in their safe route options. Without widespread smooth pavement, cyclists must ride on routes with the lowest traffic volumes – a much safer option than numbered state highways, which are major routes for regional motor vehicle travel. While less traveled routes allow cyclists to ride comfortably in the lane, where edge conditions are less hazardous and road hazards can be avoided, suitable improvements to numbered routes would demonstrate commitment to safety and sustainability principles and recognize roads as a principle state recreational resource.

While there are many construction projects identified in the NH Ten Year Transportation Plan (Ten Year Plan or TYP), preservation of the existing system is critical to maximizing the effectiveness of limited funding. While restoring pavement condition is a wholeroad investment, obviously targeted at preserving roads for motor vehicles, cyclists would benefit from a policy prioritizing preservation of the existing system.

The pavement deterioration curve demonstrates the advantage of spending for preventative maintenance. The curve is representative of a road with a design life of about twenty years. After a roadway has been constructed, a slow decline in pavement condition is observed. As the



roadway continues to age due to climate, traffic loading, etc., a much sharper decline is typical. Applying a pavement preservation treatment will generally restore the pavement condition. If treatment is delayed for another three years, a more costly rehabilitation treatment will be required to restore the pavement condition. The cost of



the rehabilitation treatment can be upwards of ten times more than the cost of the preservation treatment.

NHDOT resurfaced just over 60 miles of pavement in the Lakes Region in 2023. This includes mostly light capital paving, with some sites of major rehabilitation in Ossipee and minor rehabilitation in Bristol. The details of this plan are included in their published annual Paving Program Map.

The 2025-2034 Draft Ten Year Plan outlines potential issues for the future of pavement conditions in the state. Due to payback of NHDOTs Transportation Infrastructure Finance and Innovation Act (TIFIA) loan the department expects to lose approximately \$20 million from their repaying budget. This is estimated to drop the miles of resurfaced roadway in half from ~440 miles to ~230 miles. Without additional funding it seems likely that our region will only have about 30 miles of repaying done by the state annually starting in 2025. This will have significant impacts on roadway conditions of the regions more substantial corridors.

## **Shoulder Space**

The lack of available paved shoulder space cycling for or shoulder level space for walking are leading concerns among survey respondents. In 2009, the NH Legislature



adopted special road rules for bicyclists (NH RSA 265:143-a). The law requires drivers to caution exercise when approaching bicyclists by leaving a prudent distance between the vehicle and bicycle. A prudent distance is considered to be at least three (3) feet when the vehicle is traveling at 30 miles per hour or slower, with an additional foot of clearance for every 10 mph. After multi-use trails, rural road shoulders are also the most common places where survey respondents report walking. Availability of shoulder space is dependent upon multiple factors, including pavement width, lane width, number of lanes or striped medians, right of way width, and physical constraints. For example, West Shore Road along Newfound Lake is severely constrained by near vertical bedrock along sections on the west side of the road and by the lake on the east. Safe accommodation of pedestrians and cyclists along this popular route is a challenge that will require a creative solution.

In the Netherlands, roads like this have a configuration that accommodates all users within the existing space. For example, an 18-foot segment of pavement would have two 4-foot paved shoulder lanes, and one 10-foot center lane. The road legally accommodates two-way motor vehicle traffic – cars drive along either side of the



18

road, straddling the striped shoulder lane. When a car encounters a cyclist, it moves to the center of the road and slows down considerably. If two vehicles encounter each other in the center of the road while heading in different directions, they are moving so slowly that they are no danger to each other. Once the vehicle has passed the cyclist, the car moves back to the right and resumes at a reasonable speed.

While this traffic configuration is essentially non-existent in the United States, it is a successful example of serious pedestrian and bicycle prioritization that still manages to accommodate motor vehicles along This lower volume roads. traffic configuration is successful for three important reasons: (1) slow vehicle speeds; (2) high maneuverability of cars at no physical cost to the driver;(3) the significant physical protections for drivers traveling inside of motor vehicles. Configurations like this offer the kind of simple design solutions that gives the Netherlands the distinction as a country with the highest non-motorized mode share, as well as being a destination that draws tourists eager to cycle the country's cities and regional trails.

While the road configuration example from the Netherlands is illustrative of a creative and unorthodox approach, the reasons for its success are important considerations that should inform all transportation planning and design. Traffic calming is possible for many projects where safe separation cannot be achieved within available space, and where cultural travel norms insist on equal accommodation of travel modes.

In New Hampshire, many segments of local and state roads have lane widths that are in excess of the maximum FHWA recommended width of 12 feet. LRPC has produced a map using NHDOT data showing available road shoulder s



pace on all road segments classified as minor collectors or higher. The map, included in the Appendix, shows road shoulder space in three categories, based on available data: 0-2 feet, 3 feet, and 4 feet or greater. It also indicates road segments where paved shoulder width could be increased from the lowest category to a higher category through the striping of lane widths at 10 feet, within the American Association of State Highway Officials' (AASHTO) recommended width range for arterial routes. The analysis of the NHDOT data indicates that over 30 miles of road shoulder along state routes in the Lakes Region could be improved from the worst category (0-2 feet) to at least three feet by striping 10-foot lanes.

Many road segments also have center turning lanes that accommodate only a small number of turns at the expense of shoulder width (two travel lanes and a turning lane with minimal shoulders as opposed to no turning lane and wider shoulders). Center turning lanes should be very carefully considered, and should only be included in transportation project plans where significant delays would be likely to result from turning activity. These lanes not only consume existing pavement that could otherwise be shoulder space, but they also promote higher speeds due to reductions in visible and physical constraints within the right-of-way without necessarily increasing vehicle throughput. Such lanes can be



avoided by utilizing parallel access roads internal to adjacent development, and by accepting a certain amount of delay that does not result in overall congestion. Their visual impact on the streetscape is substantial, and they cost more to build and maintain.

## Lane Widths

The American Association of State Highway and Transportation Officials (AASHTO)"Green Book" commonly is referred to as the traffic design bible (the most recent edition is officially titled Geometric Design of Highways and Streets, 6th Edition, 2011). The Green Book calls itself a "guide" – meaning that it contains suggestions - but many states and municipalities treat AASHTO recommendations as required standards. Even so, while AASHTO publications tend to suggest ranges for key road dimensions, with the exact size to be chosen depending on the particular context of each project, professional practice is to use the maximum size whenever possible. This is partly because of the tendency to "overbuild" by designing roads to handle faster speeds, heavier traffic, and bigger vehicles than officially anticipated - partly to provide an extra margin of safety, partly to prepare for possible growth, and partly because of the "cars are the future" attitude that has dominated American planning and culture for much of the past century.

For lane widths, AASHTO suggests that "local" roads should be constructed with 9 to 12-foot lane widths and that "collectors" and "arterials" have 10 to 12-foot lane widths. The Federal Highway Administration's publication, Mitigation Strategies for Design Exceptions (July 2007, chapter 3) states: "Narrower lane widths may be chosen to manage or reduce speed and shorten crossing distances for

pedestrians. Lane widths may be adjusted to incorporate other cross sectional elements, such as medians for access control, bike lanes, on-street parking, transit stops, and landscaping. The adopted ranges for lane width in the urban, lowspeed environment normally provide adequate flexibility to achieve a desirable urban cross section without a design exception."

However, NHDOT policy is to stripe lanes at 12 feet or to the pavement edge if 12 feet cannot be accommodated with available pavement. Given the narrow and unchanging distance between buildings on many urban streets, maximum traffic lane widths often leave no room for widening sidewalks, installing bicycle lanes, center medians, or intersection bulb-outs, storm replacing sewers with more environmentally friendly stormwater treatment, adding street furniture, trees or other aesthetic improvements, and much else that would make streets more attractive and useful.

Practice of this policy is evident along many of the numbered routes in the Lakes Region pavement is too narrow where to accommodate 12-foot lanes. Striping often follows the pavement edge, yielding variable lane widths along many segments. Other sections of road have been striped at widths much greater than 12 feet. In some places this is driven in part by intersection design and other factors, and may be somewhat defensible, but wide travel lanes promote higher speeds which are exponentially associated with higher risk of pedestrian fatality.

## **Bikeability Assessment**

In 2019, LRPC performed an assessment on the "bikeability" of the region. The purpose of this report was to analyze current conditions and work being done in order to



direct future bike and pedestrian efforts by the Commission. The assessment uses the Level of Traffic Stress (LTS) as an indicator of how safe roads are for bicyclists and pedestrians. Higher LTS roads are more unsafe and require experienced users, while lower LTS are safe for users of all levels. The measurement of LTS does not include speed in this assessment because not all Lakes Region communities had this data available.

It was discovered that many of the rural backroads of Lakes Region communities had low LTS (1-2), while the larger main roads and highways had higher LTS (3-4). The figure below shows this relationship. Such a large portion of LTS 1 roads may be deceiving because they are mostly rural roads which are not always connected. This means that most residents would be able to bike and walk comfortably around their



**Figure 1**. Shows that the majority of roads in the Lakes Region are LTS 1 because there are so many rural roads. The ones with LTS 3 or 4 are the main roads and highways that act as important connections between regional hubs.



LTS Score by Functional Class

**Figure 2** breaks down the roads by type and LTS levels. The shaded area is the proportion of population that can access the roads based on LTS score. This means that very few people can safely access LTS 3 and 4 which are roads of greater importance.



neighborhoods, but would not be able to safely access significant destinations along the main corridors. In other research by Laura Getts (2017), it was determined that "97% of regional routes cannot be completed without traveling along an LTS 3 or 4 segment". This issue creates a significant break in connectivity because bicyclists and pedestrians cannot access local hubs or destinations.

One of the main areas identified in limiting connectivity is the Route 3 corridor from Franklin to Laconia. This road has a large section of LTS 4, the most dangerous condition, which limits access to a lot of resources and destinations along the corridor.

The assessment outlines priority projects to help reduce the LTS of roads in the Lakes Region. All of the projects here are included in the Ten Year Plan and are expected to be addressed by 2027. The report also outlines a simpler solution of using the planned NHDOT paving schedule to widen the shoulder widths of roads with painting rather than expanding the paved section. This will be an important tactic to follow up on and make communities aware of this opportunity to improve bike and pedestrian access on main corridors.

The assessment also recommends further analysis in order to see how connectivity can be improved throughout the region. A Network Analyst extension will be needed for a more detailed plan of which roads will provide the highest levels of connectivity and access to a variety of destinations for users. These strategies have been incorporated into the Action Plan described in Chapter 7.

## **Electric Bikes (E-Bikes)**

In E-Bikes have become a more popular form of transportation in the past few years. E-bikes can accommodate a larger portion of the population since they lower the physical requirement for bicycling. Due to the ease of use and speed of travel, this new technology comes with some considerations that should be addressed. This includes rules/regulations, trail maintenance/material, and charging infrastructure.

E-bikes range depends on usage and battery size. Having reliable charging infrastructure will be important for commuters, tourists, and recreation.

Class 1: A bicycle equipped with an electric motor, of 750 watts (1HP) or less, that provides assistance only when the rider is pedaling, and that ceases to provide assistance when the e-bike reaches 20 mph.

Class 2: A bicycle equipped with a throttleactuated electric motor, of 750 watts (1HP) or less, that may be used to provide assistance until the e-bike reaches 20 mph.

Class 3: A bicycle equipped with an electric motor of 750 watts (1HP) or less, that provides assistance only when the rider is pedaling and that it ceases to provide assistance when the e-bike reaches 28 mph. a factory installed speedometer is also required.

## NH Bike Walk Alliance (NHBWA)

E-bikes require less resistance to operate which leads to a higher portion of the population that are able to utilize this mode of transportation with ease.



## Regulations

E-bikes are under the same regulations as traditional bicycles.

- Same rules for roadway use
- Helmets required for Class 1 & Class 2 e-bikes for users under the age of 16, 18 for Class 3.
- A city, town or state agency that has jurisdiction can restrict where e-bikes are allowed for safety reasons or to protect natural resources
- E-bikes are not subject to registration, licensing, or insurance requirements
- Class 1 and Class 2 e-bikes may be ridden on bicycle or multiuse paths where bicycles are permitted.
- Class 3 e-bikes are only allowed on roadways, unless otherwise allowed by the local authority.

Federal and NH State laws consider e-bikes to be motorized vehicles. This limits which trails e-bikes are allowed to be used on.



## **Bicycle Routes and Maps**

This map shows the current bicycle routes available across the Lakes Region.





## CHAPTER 5: USE TRENDS AND PUBLIC INPUT

## **NH and US Data and Policies**

The 2020 5-Year American Community Survey (ACS) for the State of New Hampshire reveals that only 2.79% of commuters either bike or walk to work. This number is quite low in comparison with the national portion of 13% of all trips that are bike and pedestrian users. Although it is small, the New Hampshire rate remains consistent with previous trends throughout the past decade.

Research done by the League of American Bicyclists continues the work of The Alliance for Biking and Walking by investigating bicycle and pedestrian trends. The data shows that while NH has a 12% fatality rate for bicyclists and pedestrians, there has been a decrease in federal transportation funds directed towards bicycle and pedestrian uses. Just from 2017 to 2020, New Hampshire reduced its federal funding to bike and pedestrian services by 76%. This major withdrawal of funding could have a detrimental impact on improving the safety of bicycle and pedestrian infrastructure.

The New Hampshire Strategic Highway Safety Plan for 2022 outlines one of its "critical emphasis areas" as non-motorized vulnerable road users, which includes pedestrians and bicyclists. The plan includes strategies to improve the safety of these users. It largely promotes the adoption of complete streets practices throughout the state as enforced by NHDOT. The U.S. Department of

Transportation (USDOT) defines complete

#### streets as follows:

"Complete streets are designed and operated to enable safe use and support mobility for all users. Those include people of all ages and abilities, regardless of whether they are travelling as drivers, pedestrians, bicyclists, or public transportation riders. The concept of Complete Street encompasses many approaches to planning, designing, and operating roadways and rights of way with all users in mind to make the transportation network safer and more efficient."

The Federal Highway Administration (FHWA) is working towards this goal by training local agencies how to incorporate complete streets into their planning process. The safety of vulnerable users is an extremely important consideration when analyzing proposed transportation а project. LRPC also aims to address this concern and prioritize complete streets practices by designing a safety action plan in the 2023 Regional Transportation Plan. This plan will help guide communities in their project design process with an emphasis on safety and accessibility for all transportation users.

## **Concerns Expressed Through Public Meetings and Survey**

The summary need/concern statements that were generated through the public meeting process are included here. Many of these statements represent expressions of



concern that were shared by multiple people at all five meetings. The large set of summary statements provided below, helped to define five broad topical areas to target planning recommendations – Design & Function, Connectivity & Opportunities, Information & Planning, Education & Enforcement, and Funding & Policy.

## **Information and Plans**

- Economic benefits of trails and walkable communities are poorly understood.
- People are unaware of existing trail and walking resources.
- Communication with residents about NH DOT paving schedule is poor.
- Signage indicating safe routes through regional and village centers for road cycling is poo.
- Decision makers need to be better informed about resident priorities and economic benefits of walkable and bikeable communities.
- 6) Untapped local knowledge could help inform route identification and map development.
- Selectmen and road agents need information about right-of-way designs that can calm traffic and that will be approved by NHDOT.
- 8) information on what NHDOT is responsible for on state roads.
- Need for examples of model local pedestrian and bicycle plans for reference.
- 10) Need for useful regional maps that show gradation of safe routes to unsafe.
- 11) Need for regional walking resource/destination map.



Fig. 2 - Sample of notes from public meetings

12) Need to identify the most dangerous locations/conditions in all thirty-one communities.

## **Funding & Policy**

- 1) New sources of funding for projects need to be identified.
- Municipalities' and NHDOT refusal to agree to maintain new proposed sidewalks is leading to their removal from concept plans.
- Snow clearing on sidewalks is poor and unsafe for pedestrians and wheelchairs in most places.
- 4) Limited funding due to misproportioned/misaligned allocation federal and state allocation.
- 5) Maintenance inconsistencies exist between rural areas and Urban Compact Districts.





- 6) Competing needs for same pool of funds.
- 7) Need for local comprehensive plans as well as regional plans.

## **Design & Function**

- Cycling space is needed for circumlake routes that are popular and that could draw more visitation and/or events.
- Transit routes need shoulder space for safety of demand response service.
- 3) Paths and walking routes to schools need adequate lighting.
- 4) Existing sidewalk infrastructure does not meet ADA access in urban areas for passing and safe crossing in wheelchairs. Designs are haphazard and inconsistent, with huge curb cuts, and dangerous out-slopes and tip-downs at intersections.
- 5) Summer parking demand is creating traffic congestion and hazards for bikes and pedestrians.
- 6) Little use of traffic calming measures anywhere in the region.
- Recreational events require more police and road closures to mitigate unsafe conditions.
- Some routes have severe space constraints and no reasonable alternative route for miles due to geography.
- Intersections are designed for cars and do not minimize pedestrian hazards.
- 10) No signed on-street bike lanes anywhere in the region.
- 11) Travel lane space allocation created by 12-foot or greater lanes promotes excessive speeds.
- 12) Pavement conditions are poor for riding in many places.
- 13) Crosswalks have poor line-of-sight and have unsafe landings in many places.

- 14) Parking stalls have been striped too close to crosswalks and intersections in violation of state law in many communities.
- 15) Observed speeds do not align with road conditions and settlement characteristics.
- 16) Lack of adequate (or maximized) shoulder space due to pavement width or striping.
- 17) Plastic road paint is a hazard to cyclists when wet.

## **Education/Enforcement**

- Share the Road signs needed along narrow and busy corridors, but sign requests are evaluated by NHDOT based on road design/function criteria rather than on informational needs.
- School-based riding and walking trainings for children and adults are needed.
- Landowners need to be educated on the benefits of expanded rightof-way to accommodate bicycles and pedestrians.
- 4) Personal safety on trails is a concern for some.
- 5) Visitors are surprised by bicycles and pedestrians in the road.
- Tourists drive aggressively, trucks and commercial drivers often pass too close, and other drivers are fearful of cyclists and pass too widely.
- 7) Wood chip trucks present a hazard for road cyclists.
- 8) Pedestrians and bikes often use the wrong side of the road.

## **Connectivity/Opportunities**

1) Need more transit options to increase mobility.







- 2) Connectivity between existing multiuse trails is limited.
- Need to identify low-cost, feasible connections as opposed to shortest route infrastructure.
- Connectivity to outlying areas is poor but people walk along nonexistent road shoulders all the time because it is the only place for them.
- 5) Not enough family friendly bike routes through communities.
- Connections between trails and transit do not exist but could be made.
- Seniors and children are not walking much to schools and within the community but say they would if conditions were safer and if it were possible.
- Lack of safe pedestrian connections between neighborhoods and popular destinations.
- 9) Poor connectivity of existing sidewalks and trail systems.
- 10) Need for additional and improved off-street multi-use trails to support economic development as well as travel options.

## **Bicycle and Pedestrian Data** Collection

A comprehensive assessment of existing conditions cannot be limited to the built environment alone, but must include recent usage trends, stated needs, and established goals. Together with the built environment, these factors help to illustrate current conditions affecting bicycling and pedestrian accessibility in the Lakes Region. As mentioned in Chapter 3, the LRPC conducted a survey in the winter of 2022 about the bicycle and pedestrian patterns in the region. The following conclusions have been made from the results of this survey:

## **Respondent Data**

- 634 responses were received, which is over double that of the 2011 survey.
- 79% of respondents were 50 years or older, a 12% increase from 2011.
- About half of the respondents were from Laconia, Hebron, Gilford, or Meredith.
- 69% of respondents believe that local governments should be responsible for improving bicycle and pedestrian infrastructure.



## Walking Data



**46%** of respondents believe that there are places in their community in need of repaired or new sidewalks









Other 12%

47%

300

400

#### Shoulders of Numbered Highways 9% Sidewalks 3% Top 3 Reasons for Bicycling Rural Town Roads 40% Off Street Paths/Trails 36% For Exercise To Be Outdoors 2 To Spend Time Where do you primarily bike? ŝ with Others The leading factors that Most bikers ride on either rural town roads or prevent respondents from off-street paths and trails biking are: • Traffic safety concerns What **prevents** you from biking • Lack of bike lanes more or at all? • Weather **Traffic Safety Concerns** Lack of Bike Lanes 35% Weather 26% The main improvements that Lack of Trails & Off Road Bike 25% would encourage Paths respondents to bike more Pavement Safety Concerns 20% are: Narrow Road Shoulders 17% o 4ft. shoulders on Time Constraints 15% paved roads Don't Own a Bike 7% • More off-road bike Lack of Motivation paths and trails 5% • Dedicated on-street Other 5% bike lanes 100 200 0

81% of respondents would like more off-street paths and trails in their community.



**Bicycling Data** 





## CHAPTER 6: BIKE AND PEDESTRIAN PROJECTS

## **Overview**

The Lakes Region has implemented many projects to improve bike and pedestrian infrastructure since the last version of this report in 2012. This chapter includes three major projects that have altered the way and bike within their people walk communities. These projects have improved the accessibility and or safety of vulnerable road users. Each project is outlined in detail and considered for its recent effects on the community. This analysis is done with the hopes of improving future bike and pedestrian projects in the region so that communities can continue to grow the infrastructure available to bicyclists and pedestrians and bring the 2023 vision to life.

## **Recently Completed Projects**

#### **Bristol Downtown Improvements**

One of the major pedestrian-focused projects undertaken by a community was the complete renovation of the Bristol downtown area of Central Square. Bristol's Master Plan outlines the need for certain pedestrian safety and accessibility features that would improve the vitality of their downtown region. The proposed projects included adding curbing and other buffer options such as widening the sidewalks or adding trees and décor that help separate the pedestrians from motor vehicles to reduce their vulnerability. Bristol has also created the new Bristol Falls Park. It includes a handicapped-accessible entrance right by Central Square. This allows pedestrians of all physical abilities to participate in the use of this trail and encourage community recreation. Bristol also hopes to expand this trail along the Pemigewasset in the upcoming years with the creation of the Pemi – Path.

The Pemi – Path would be a 2.5 mile, multiuse path that would connect to the Northern Rail Trail through other multi-use trails. By making the downtown area a safe and accessible place for pedestrians, Bristol aims to increase participation in various economic and recreational activities and promote a walkable community lifestyle.







communities involved are prioritizing these modes of transportation by electing to invest in an extension of the popular Phase I construction of the trail. This project also demonstrates how successful it can be to build off existing infrastructure. The WOW trail has proven to be well used and appreciated by the communities. The 2016 extension furthered this use and opened up trail accessibility to more citizens. The WOW organization has an ultimate goal of continuing extensions so that Weirs Beach will be connected down to Franklin by one cohesive rail trail.

## **WOW Trail Extensions**

After its Phase I construction in 2010, the WOW trail underwent further extension in 2016. The Phase II project consists of a 1mile long, shared-use, rail-with-trail path from the Belmont Town Line to the south end of the completed Phase I path segment. Similar to Phase I, a section along Fair Street was constructed on-road to avoid needing to build a pedestrian bridge over the Winnipesaukee River. In addition. another section of path was constructed to provide full access to the City-owned Bartlett Beach facility. The path is 10 feet wide and paved with an attractive bridge over Durkee Brook and woodconstructed boardwalks over sensitive environmental areas. The northern segment begins at the former train station where the project included a full-width concrete paver surface with granite block inlay and new aesthetic lighting. From the train station, the path continues on-road along New Salem Street. Construction of this Phase II was completed in 2016.

This project serves as an example of continued dedication to bike and pedestrian accessibility and safety. The



The WOW trail is shown above where the Phase II addition begins. Here, users can access Bartlett Beach or continue along the extended trail into Belmont.



#### **New Cross Walk in Meredith NH**

In 2019, a cross walk was added to downtown Meredith to help pedestrians safely cross US 3. It is located in a very high traffic area on the state highway and serves to connect the Main Street part of town with the Town Docks and sculpture walk. This crosswalk has extra safety features to



accommodate such a dangerous area for pedestrians. It is known as a HAWK beacon (High-Intensity Activated crosswalk), a traffic signal used to stop road traffic and allow pedestrians to cross safely. It is more commonly known as a "pedestrian hybrid beacon". The purpose of a HAWK beacon is to allow protected pedestrian crossings, stopping road traffic only as needed. It is designed with flashing lights in order to increase awareness of pedestrians and reduce their risk of danger. However, there have been many public concerns about the efficiency of this design due to confusion about its operation. Many drivers and pedestrians are unfamiliar with this design and are unsure of the right of way here and the meaning of the different light signals. This confusion can ultimately put more pedestrians at risk if users, both drivers and pedestrians, are not sure how to proceed at this crosswalk.



Photo courtesy of The Laconia Daily Sun.



## **Upcoming Projects**

The 2023-2032 NHDOT Ten Year Plan outlines the transportation projects that will take place in the Lakes Region throughout the next decade. The existing bike and pedestrian projects are shown in the table below along with which of the goals from this plan they serve to address. This will help the LRPC recognize which of the bike and pedestrian goals are being worked towards and which ones may need more attention in future projects and plans.

Municipality	Project #	Scope	Construction	Priority	Goals
			Dates		Addressed
Belmont	40635	Improve intersection safety and congestion	2025		Safety, Design,
Bristol	40636	Roadway widening and shoulders for bike/ped travel from School St. to West of Danforth Brook Rd.	2026		Safety, Connectivity, Economy and Health
Bristol	41579	Bike/ped improvements to Lake St., increase separation between different modes of transportation for safety	2026		Safety, Design, Economic and Health (ERZ)
Franklin	42513	Construct a pedestrian walkway on the existing trestle bridge to connect to Mill City Park	2026		Safety, Connectivity, Equity
Gilmanton	42603	Construct pedestrian islands and sidewalks and improve curb at intersection of NH 140 and NH 107	2030		Safety, Design, Connectivity, Economy and Health
Laconia	43731	Construct 4,300 ft. of sidewalk on Elm St. and 8' wide multi-use path on the other side of Elm St.	2031		Safety, Design, Connectivity, Equity, Economy and Health
Laconia	43845	To replace and improve a municipal red-listed bridge on US 3 / Weirs Blvd., improve sidewalks and bike/ped safety	2026		Safety, Design, Economy and Health
Meredith	43533	Intersection improvements for Laker Lane, True Rd., Quarry Rd., and Patrician Shore Circle with NH 25	2032		Safety, Design, Connectivity
Moultonborough	40639	Intersection improvements on NH 25 between Lake Shore Drive E and Lake Shore Drive (W)	2025		Safety, Design
Moultonborough	41580	Complete streets improvements to Moultonborough Central Village	2027		Safety, Design, Connectivity, Economy and Health, Equity
Moultonborough	41581	Intersection improvements at Sheridan Rd. and NH 25	2027		Safety, Design







Plymouth	41583	Intersection and sidewalk improvements to Highland St. and in the vicinity of Reservoir Rd.	2025	S	afety, Design
Plymouth	43532	Intersection improvement to the NH 25/ Smith Bridge Rd.	2032	S D C E H	Safety, Design, Connectivity, Economy and Health
Tilton	42600	Intersection safety improvements (roundabout)	2031	S D C E H	Safety, Design, Connectivity, Economy and Health
Wolfeboro	29615	Improvements from NH 109 to Alton TL, consider roundabouts, pavement width, sidewalks, work with SRTS to consider safety issues for pedestrians as this road includes access to 4 schools and other public facilities	2025	S D C	Safety, Policy, Design, Connectivity

Many of these upcoming projects address multiple goals from Chapter 2 of this report. This shows how many aspects of bike and pedestrian infrastructure are intertwined so that by improving one aspect, it may also benefit another. For example, by improving the safety of biking and walking on a roadway, it automatically increases accessibility for users because of the lower LTS and contributes to connectivity throughout the region. It is also important to note that all of the projects above work specifically towards the goals of safety and design for bike and pedestrian infrastructure. The Lakes Region communities and the NHDOT prioritize safety and altering design elements to improve it in the long run. By analyzing these projects in terms of the goals of this report, it shows that all the work being done is actively contributing to the LRPC's 2023 vision, often in multiple ways through each project.



## **Project Recommendations**

The Lakes Region Planning Commission has identified multiple bike and pedestrian projects that will greatly benefit the infrastructure across the region. These projects were selected based on the goals from this report that they will address, the needs of the community, and feasibility considerations with the funding and resources available. LRPC recommends that these projects should be considered as priorities for the upcoming Ten Year Plan, or other funding sources. The highlights below detail how each project exemplifies the goals of this Bike and Pedestrian Plan and will help bring to life the vision of a walkable and bikeable Lakes Region.

## Plymouth North Main Street Safety Improvements

The Town of Plymouth proposed a project for the improvement of US-3 (Main Street) to be included in the 2025-2034 NHDOT Ten Year Plan. The proposal includes a correction of horizontal and vertical road alignments to improve sightlines, as well as adding in pedestrian infrastructure such as sidewalks, curb ramps, and crosswalks along the project area spanning along US 3 from Tobey Road to Fairgrounds Road. The project areas and major destinations are shown in the photo further below.

The Town expressed great need for this project because it was determined to be an extremely unsafe area for both pedestrians and drivers. Crash data from the Plymouth Police Department has shown a recent increase in frequency of crashes at the Amory Road intersection. In fact, the proposal states that this section of the US 3 road is the 2<sup>nd</sup> greatest safety challenge area and the only one yet to be addressed. The Town notes that there has been a lot of interest in this project due to the high volume of traffic in this section from the NH Army National Guard Amory, Plymouth State University parking, and the nearby downtown area. The 2018 Plymouth Master Plan (see Appendix C) supports the construction of safe bike and pedestrian mobility, which will be accomplished through this project.

The benefits of this project along Main Street in Plymouth address the goals of Design, Safety, Connectivity, Equity, and Economy and Health.

**Design** – The Town's planners, LRPC, and NHDOT are currently working together throughout the planning and design process to incorporate sufficient bike and pedestrian elements to reduce the vulnerability of users.

**Safety** – One of the major motives of this project proposal is to improve the safety of both pedestrians and drivers by altering the street design and elements. The goal is to create safer conditions for all road users with these improvements.

Connectivity – The stretch of US 3 from Tobey Road to Fairgrounds Road offers access to many different types of destinations including Plymouth State University, the National Guard Amory, the downtown area, and the development at Foster Street with the Common Man Inn and Foster's Boiler Room Restaurant. By improving pedestrian access to these destinations, Plymouth would greatly increase the connectivity for bicyclists and pedestrians between more rural areas and the vibrant downtown.



Economy and Health – The accessibility to the destinations mentioned above will also promote economic development because more transportation users have access to these stores and restaurants. It allows bicyclists and pedestrians to use these modes of transit for everyday actions such as errands and social outings. Hopefully, this will encourage people to use these modes of active transportation more often, which will eventually improve public health as well.

Equity – In 2019, Plymouth was designated as an opportunity zone by the US Department of Transportation. These zones "are economically distressed communities, defined bv individual census tract. nominated by America's governors, and certified by the U.S. Secretary of the Treasury" (USDOT). Since the proposed project is meant to help promote local economic activity, it will be supporting a Lakes Region community in need. This project demonstrates how bike and pedestrian infrastructure can be used to

address issues of transportation and economic inequalities among communities.

The LRPC Transportation Advisory Committee met in the spring of 2023 to rank all of the proposed projects based off of a number of criteria including safety, mobility, equity, and economic development. The Plymouth project scored 3<sup>rd</sup> ranked priority for the 2025-2034 Ten Year Plan, only behind the second ranked by 0.08 of a point. Although it scored highly among the TAC, the Plymouth Main Street project was too expensive to fund with the allocated TYP amount from NHDOT.LRPC therefore recommends that this project be considered in future years when the funding is available because it provides many contributions towards the 2023 vision of this report. This project will be very important to the future of downtown Plymouth in both its physical and economic development.



#### **Franklin Trestle Bridge**

The City of Franklin proposed a project to create a pedestrian walkway on the Trestle Bridge to connect two recreational areas. The bridge is shown in the photo below with Mill City Park to the left and the Winnipesaukee River Trail entrance to the right. It is still being decided what type of construction will occur, whether the City can salvage and restore the original bridge or if a total reconstruction is needed. Either way, this project will greatly benefit the community of Franklin and will address several goals of the 2023 Bike and Pedestrian Plan. transportation users by removing them from the roadway. This project is currently a planned TAP project, but there has not been enough funding to move forward to the next phase.

**Connectivity** – As mentioned before, the Trestle Bridge would connect two major recreational areas in downtown Franklin. The Winnipesaukee River Trail offers miles of safe trails for cyclists and pedestrians. Mill City Park is a developing public area with lots of access to nature and outdoor space. The Trestle Bridge would create immediate access between the two and hopefully encourage dual use between the



two sites.

**Safety** – The location of this trestle bridge would provide pedestrians and cyclists with an alternative to the busy highway road. The bridge would connect the Winnipesaukee River Trail with the Mill City Park, both near downtown Franklin along Central Street (US 3/NH 11). Pedestrians currently have to walk along the side of Central Street in order to access Mill City Park from the rail trail entrance. This section of the road is labeled by NHDOT as a level 4 traffic stress area and is extremely dangerous for bike and pedestrian use. The construction of the trestle bridge would completely eliminate this risk and reduce the overall vulnerability of active

Equity – Franklin has been identified by the US Department of Transportation as a highly "transportation insecure" city using their Equitable Community Transportation Explorer program. The greatest challenges Franklin are traffic safety in and transportation access. The Trestle Bridge project will help to improve the safety of vulnerable transportation users and also improve access to active modes of transportation. It will work to address the issues of inequality within local transportation.



Economy and Health – The City hopes to increase use of the rail trail and park with the addition of the trestle bridge walkway. The increased access to both recreational facilities will likely increase the number of people who visit them. This, in turn, will promote better public health by encouraging outdoor activities and active modes of transportation. Due to the location of this project being so near to the recently revitalized Franklin downtown area, it also serves as a way to increase economic development. One suggestion to further this impact would be to ensure safe crossing of Central Street for pedestrians to access the shops downtown.

This project was first proposed in 2018 and included in both the 2021-2030 and 2023-3032 TYP's. In 2019, the City of Franklin received a \$500,000 grant from NHDOT for this project. However, the project still requires more funds in order to be completed. LRPC recommends that this project be considered in future reviews for TYP funding because it aligns well with the 2023 vision. This project will help make Franklin a walkable community and promote recreational activities, just as the City advocated for in its 2018 master plan (Appendix C). As of the most recent TYP report, it will require about \$700,000 more for completion.

## **Alton-Gilford NH 11 Renovations**

Back in 2015, the Towns of Alton and Gilford submitted their first project proposal for the rehabilitation of NH Route 11. This led to a study being conducted from 2019 to 2022 in order to assess the needs of this corridor. The conclusion from this study includes the following recommendations:

- Intersection improvements of NH Route 11 near Ellacoya State Park and at NH Route 11A, both with roundabouts recommended for traffic calming measures.
- Widen feasible sections of the NH 11 corridor ranging from Ellacoya State Park to Minge Cove Road. This is recommended to be done in segments due to the high funding requirements.

Based on the findings of this study, LRPC supports the continuation of this project and recommends it for further consideration in future Ten Year Plans. This project will largely address the goals of Safety, Connectivity, and Data.

**Safety** – One of the main objectives of this project is to widen NH Route 11 wherever possible to 11' travel lanes and 5' shoulders. This would provide cyclists and pedestrians with more room to safely navigate across the corridor and would decrease the level of traffic stress that they The experience. roundabout recommendations work as safety measures for both drivers and pedestrians; they reduce the vulnerability of those crossing the street and act as a traffic calming measure to reduce speed on the busy highway.

**Connectivity** – NH Route 11 is a major transportation corridor in the Lakes Region that connects the town centers of Gilford and Alton, while also providing access to recreational sites such as Ellacoya State Park and the Mount Major trailhead. This means that this would be a major step towards connecting the Lakes Region through bikeable and walkable infrastructure.



**Data** - This project has already been extremely informative in safety analysis data, thanks to the completed planning study. It has given the LRPC a formative base to make decisions about the necessity of bike and pedestrian infrastructure in this area.

This project will transform the long-range active transportation opportunities available to the Lakes Region. It has already contributed to the goals of this plan and will continue to help achieve the 2023 Vision. The LRPC supports the continuation of this renovation of NH Route 11 in Alton and Gilford.







# CHAPTER 7: ACTION PLAN TO SUPPORT THE VISION

A purposefully connected network of trails, sidewalks, road shoulders, and markings promoting safe, enjoyable, and inclusive bicycle and pedestrian mobility for residents of all ages and physical ability. Design and maintenance of livable, complete streets that support transportation, recreation, health, and economic interests throughout the Lakes Region.

The following strategies will help guide Lakes Region Planning Commission's efforts to support the vision for livable streets and walkable communities. In order foster active transportation to and recreation in the Lakes Region and make substantial progress toward achieving the 2023 vision. state and municipal governments will need to commit to a of purposeful design culture and maintenance of streets that prioritizes safety for all modes. State agency leaders

will also need to acknowledge that our streets and rail corridors are some of our most important and valuable recreational resources. They will need to collaborate and demonstrate that in addition to transportation, these assets can support statewide goals related to physical and mental health, public safety, and economic development. With state, regional, and local commitment to livable streets and walkable communities, the vision can be achieved.

## Goals:



## **Strategies and Actions:**

66

Strategies	Goals Addressed	Time Frame	Priority
Help communities articulate active transportation/recreation goals in master planning and identify opportunities to amend local road standards, land use regulations, and zoning to foster appropriate bicycle and pedestrian connections.	2 3		
Explore and promote extending current sidewalk networks, and/or creation of new sidewalks, in areas that currently have compatible land-use regulations.	2		
Hold communities and their proposed transportation projects to design standards that comply with the Complete Streets practices in order to make roads inclusive and safe for all modes of transportation.	2 3		
Support community efforts to request lane allocation modifications (e.g. striping narrower lanes, etc.) from NHDOT on state roads in order to calm traffic speeds and provide wider shoulders for bicycles and pedestrians.	2		
Assist communities in capitalizing on the NHDOT paving schedule to alter lane paintings for wider road shoulders rather than extending the pavement in areas with limited space.	2		
Continue to explore opportunities for "Share the Road" signs and other informational aids that help publicize reasonable clearance distances between vehicles and bicycles.	6		
Continue to assist with development of Transportation Enhancement grant applications, Highway Safety Improvement Planning requests, and development of Safe Routes to School travel plans.	3 6		
Encourage and assist communities with participating in the Safe Streets and Roads for All Grant Program by providing them with an applicable safety action plan.	3		
Work with the Transportation Advisory Committee (TAC) to prioritize active transportation projects in underserved or disadvantaged communities.	4		
Continue to review and update regional bicycling and pedestrian plans on a regular basis.	2 3		
Develop a process to disseminate useful bicycling and walking information and data on a regular basis. Examples may include a web component created and maintained for this purpose.	5		
Pursue opportunities to develop an enhanced regional bicycle map showing useful information (related to conditions and safety, locations of bike shops, and completed and proposed sections of multi-use pathways) that can be distributed or made available to regional economic development and tourism organizations, municipalities, and other stakeholders.	5 7		
Promote walking, cycling, and complete streets in local context sensitive solutions for design and re-construction of state and local routes	2		







Provide assistance with pedestrian and bicycle data collection on off- street trails and paths throughout the region.	5	
Encourage local and state officials to maintain a database of regional pedestrian and bicycle collision related injuries and fatalities	5	
Encourage municipalities to study the impacts of peak-parking demand to help communities more effectively manage parking through the promotion of techniques including shared parking allocation and maximum parking limits. One goal of this effort will be to identify underutilized parking spaces for potential re-use as pedestrian and bicycle connections.	1 2	
Support a state policy of creating usable shoulder width for bicycles and pedestrians through reduced lane widths and increased pavement widths along priority routes.	3	
Support efforts to connect off-road pathways with on-road bicycle and pedestrian routes.	1	
Provide trail developers with corridor/route assessment assistance and information.	5	
Encourage Brownfields assessment work at sites with potential to enhance existing pedestrian and bicycle accessibility.	2	
Support regional transit providers and the Regional Coordinating Councils with bicycling and walking data and analysis to enhance transit service planning.	5	
Coordinate with Lakes Region communities for increased participation in Bike/Walk to Work Week, and other programs that encourage active transportation.	3 7	
Encourage NHDOT to support an economic impacts study of bikeable/walkable communities in order to support community decisions about constructing and maintaining infrastructure	7	
Integrate active recreation in the Scenic Byway Corridor Management Plan.	3	
Continue to participate in the Safe Routes to School State Advisory Committee.	3 6	
Promote and participate in educating the public on the value of active communities and healthy lives.	7	
Communicate with railroad owners to identify opportunities for future trails.	1 2	
Support an effort to work with cyclists familiar to the region, bike shop and chamber of commerce representatives, and other stakeholders to identify safe existing routes, and priority routes of regional importance (triathlon routes, long distance bike race routes, and circum-lake routes, for example) to be included with future maps	2	
Work with the Federal delegation, State legislators, and NHDOT staff to preserve or increase funding for NHDOT's Transportation Improvement Program.	3	



### Appendix A:

## Historical Bike and Pedestrian Documents in the Lakes Region

### Lakes Region, NH Bikeway System: Background Report (1982):

In April 1982 the Lakes Region Planning Commission (LRPC), working in cooperation with the city of Laconia Planning Department, completed the *Lakes Region, NH Bikeway System: Background Report*. Funded by the US Department of Transportation, Federal Highway Administration, this report examined the opportunity to develop a 27-mile bikeway through 6 communities – Meredith, Laconia, Belmont, Northfield, Tilton, and Franklin. The purpose of the bikeway was to provide a safe and convenient alternative to automobile transportation, while at the same time providing economic benefits to the Lakes Region in the form of tourism and recreation. Although only covering a small portion of the area, this was arguably the Lakes Region's first foray into regional bicycle and pedestrian planning and serves as a major foundation for this 2011 planning effort, 29 years later.

#### Vacation Travel Background Report: Lakes Region, New Hampshire (1983):

In 1983, the LRPC released a comprehensive report that examined the existing conditions, challenges and opportunities, and recommendations for future action regarding travel and tourism in the Lakes Region. The report established specific long-term goals for the region, which included the preservation of scenic and rural qualities, the promotion of historic and cultural activities, continued promotion of tourism, recreation, and improvements of resort, commercial, and downtown facilities. The report also examined the barriers and challenges to transportation within the region, recognizing that increased traffic congestion would have negative impacts on the travel and tourism industry, and recommended that the use of bicycles and bicycle touring through the Region be encouraged.

Of perhaps greater significance, however, was the attention the report gave to linking the major natural resources of the region (scenic views, lakes, rivers, etc.) with future tourist development. Specifically, the report recommended that recreational activities such as bicycling and walking be encouraged to provide residents and visitors both "exercise and the quiet enjoyment of the region's natural heritage". The report identified that adequate bicycle and pedestrian facilities were sorely lacking throughout the region.

Some of the major recommendations of the report related to bicycle and pedestrian planning included the following:

- Encourage communities to reinvest local tax revenues generated from tourism in the construction of recreation facilities such as jogging and bicycling trails.
- Recognize that tourists use several modes of travel while visiting the region, and to expand the transportation network to include roads and highways, sidewalks, pathways, railroad tracks, boating facilities, and airport facilities.
- Make public improvements to bicycle and pedestrian travel. Pedestrian improvements could include the construction or widening of sidewalks, trees or other plantings, benches, curb ramps, crosswalks, signals, and signage. Bicycle improvements could include bicycle racks, road shoulder improvements, signage, and the creation of bike routes and pathways.



 Recognize that downtowns and village centers play an integral role in the tourism industry in the Lakes Region, and that efforts need to be made to enhance pedestrian circulation so that after finding a place to park, tourists can walk to the businesses they intend to visit. Barriers should be removed, crosswalks should be safe, pathways should be well landscaped, and benches should be provided for both residents and tourists of the Lakes Region.

#### Lakes Region Transportation: Baseline Report (1986):

In 1985, LRPC, responding to a request by the New Hampshire Department of Public Works and Highways (now Department of Transportation), agreed to develop a comprehensive Lakes Region Transportation Program. LRPC assumed the role of coordinating and furthering the activities of a newly created Regional Transportation Committee (now referred to as the Transportation Technical Advisory Committee or TAC). The first responsibility of the Regional Transportation Committee was to develop goals and objectives for the new Lakes Region Transportation Program. From those goals and objectives, the committee established specific transportation-related tasks to be undertaken in the program. One of the early tasks, funded through a grant from the State, was for the committee and LRPC Staff to collect data and develop an inventory of the existing regional transportation network. Because the ultimate purpose was to develop a comprehensive transportation plan for the Lakes Region, the focus of the data collection and inventory was on streets and highways, bridges, railroads, airports, and public transportation.

The body of the *Lakes Region Transportation: Baseline Report* presented no information regarding bicycle or pedestrian transportation. However, an appendix to that report contained a list of the goals and objectives originally identified for the first ever Lakes Region Transportation Program, which included the following:

- To encourage and support a multi-modal transportation system in the Lakes Region by:
  - Assisting and encouraging pedestrian movement with a sidewalk construction program with crosswalks in highly developed areas.
  - Encouraging the development and maintenance of transportation related facilities for recreation such as bikeways and jogging trails.
- To give special attention to the natural environment and scenic qualities of the Lakes Region when considering transportation improvements by:
  - Promoting accessibility by a variety of modes to existing and future recreational facilities and open space in a way that enhances recreational pursuits.
- To ensure the transportation system is compatible with energy conservation programs, clean air standards, and the enhancement of environmental quality and safety by:
  - Encouraging the development and use of safe bicycle facilities.

#### A Regional Transportation Plan for the Lakes Region (1991):

The first Regional Transportation Plan for the Lakes Region was released in 1991 and built upon the data collected and presented in the 1986 Baseline Report. The 1991 plan reaffirmed the goals and objectives contained in the 1986 document, with only one minor change that was unrelated to bicycle and



pedestrian planning. The 1991 plan was developed to continue the process to create a comprehensive transportation plan for the Lakes Region. The focus, however, remained on roads and highways, and bicycle and pedestrian transportation was again not integrated in the planning process.

#### Lakes Region Tourism Profile (1995 and 2002):

In 1995 the LRPC released a profile of tourism in the region which would supersede the 1983 Background Report discussed earlier. This report was a significantly scaled back report, primarily focused on presenting data and existing trends in tourism throughout the nation, but most specifically in the Lakes Region of New Hampshire. As recognized in the 1983 report, the lakes and the natural resources of the region are the primary reason for the strength in the tourism industry. The report also recognized the potential for heritage-and history-based tourism to play a larger role, and that biking and walking could play a part. For example, bicycle tours connecting historic places, or local heritage walking tours could all play a part in promoting tourism in the Lakes Region in the future.

The 1995 Tourism Profile also discusses the current trends in transportation in the state and region. Unfortunately, the trends discussed are specifically focused on the automobile and its relationship to highways and roads and goes as far as to indicate that "Once in the Lakes Region, a number of connecting highways offer nearly unimpeded automobile travel within and between the many municipalities and villages". So again, although bicycle and pedestrian transportation is recognized as an important opportunity in the Lakes Region, little is flushed out regarding how future facilities and improvements should be planned. The 1995 report was updated with new information and data in 2002 and remained much the same as its 1995 predecessor.

#### A Regional Transportation Plan for the Lakes Region Update (1997):

In 1997 the LRPC developed and approved the first transportation plan for the Lakes Region that fell under the auspices of the Intermodal Surface Transportation Efficiency Act (ISTEA) legislation which authorized federal transportation programs for fiscal years 1992 through 1997. This important legislation provided the focus on future transportation efforts at the federal and state levels by providing for "intermodal" transportation which purposefully included non-motorized transportation such as bicycle and pedestrian.

The 1997 LRPC plan recognized that the automobile was still the predominant mode of transportation, and that highways were the principal, almost exclusive, means of travel in and around the Lakes Region. However, the 1997 plan represented for the first time an emphasis on transitioning the region towards a more multi-modal transportation system without a near total dependence on privately operated motorized vehicles. Intermodal planning to include bicycle and pedestrian transportation became a clear goal of the LRPC as articulated in the 1997 Regional Transportation Plan, although the plan provided significantly more focus on bicycle, as opposed to pedestrian transportation.

Of perhaps most notable importance was the Plan's recognition that transportation planning needed to be integrated with local and regional economic development, housing, environmental, historic, and land use planning. It was noted that bicycle facilities can support this integrated approach to planning. As



such, the plan attempted to identify the existing conditions in the region related to bicycle infrastructure. Based on an examination of existing and planned on-road bike routes and off-road multi-use pathways, the plan identified recommendations and strategies to improve the bicycle transportation infrastructure in the future. These recommendations were based on the identified major benefits of bicycle transportation, including:

- The reduction of motorized vehicle traffic on the highway system.
- The reduction in air pollution as a result of reduced motor vehicle travel.
- Benefits to the tourism and recreation industry in the Lakes Region.
- Other quality of life benefits to the residents of the region.

The 1997 plan concludes by identifying three specific strategies, and four recommendations related to bicycle and pedestrian transportation in the Lakes Region:

## Strategies:

- Perform community out-reach, data collecting, and research to produce a comprehensive bikeway and pedestrian map for the next updated Regional Transportation Plan.
- Review proposed projects and comment on those locations that are considered existing or proposed bike or sidewalk locations.
- Provide technical assistance to communities or civic groups considering multi-use trails.

## Recommendations:

- The bikeway system (map V-2 in the 1997 plan) will be completed prior to the year 2017
- All enhancement projects currently in the NHDOT's Ten Year Plan will remain at their assigned construction date.
- All those multi-use trails shown on map V-2 and defined as being planned, designed, or constructed by active groups will be completed by the year 2007.
- All highways as shown on map V-2 as having wide shoulders will be signed and striped for bicycle safety by the year 2007.

## Lakes Region Tour Scenic Byway Corridor Management Plan (1999):

In 1974 the State Scenic Byway system identified 67 miles of roadway, through ten communities in the Lakes Region Planning Commission's area, as the Lakes Tour. The underlying purpose of the *Lakes Tour Scenic Byway* is to promote the enjoyment of its scenic, natural, historic, cultural, and recreational attributes. In 1997, LRPC began work on a management plan for the Lakes Region Scenic Byway Tour. The overall goals of the management plan emerged as follows:

- To enhance and protect the livability of the corridor.
- To protect the unique character of all the individual communities.
- To ensure that the scenic, environmental and cultural qualities of the Corridor are protected, enhanced and managed appropriately in the future.
- To identify traditional and innovative strategies that reduces traffic on the roadways of the Byway.
- To improve the multi-modal and inter-modal considerations.



- To increase awareness of the cultural, environmental, and historical attributes of the corridor for the residents and the visitors.
- To promote sustainable low impact tourism with an emphasis on cultural, environmental, and heritage tours.

All of the above goals have a direct relationship to both bicycle and pedestrian transportation, and the contents of the *Lakes Tour Management Plan* emphasize this relationship. Chapter 5 of the Management Plan is dedicated to transportation, and discusses such issues as roadway shoulders, traffic calming, and an assessment of bicycle and pedestrian transportation. In fact, the management plan states that "bicycle and pedestrian use and safety are at the heart of good byway planning". It can be argued that bicycle and pedestrian use and safety is at the heart of all transportation planning efforts, and this understanding led to the development of this first-ever bicycle and pedestrian plan for the Lakes Region.

## PLAN 2000: A Lakes Region Transportation Plan Update for the Year 2000 (2000)

"To maintain, enhance, and manage infrastructure that facilitates, encourages, and supports viable pedestrian and bicycling transportation."

*PLAN 2000* is the most recent update to the 1997 Regional Transportation Plan for the Lakes Region. The focus of this plan is "the provision of efficient multi-modal and inter-modal transportation systems which do not negatively impact the surrounding natural, social or historic environments." The plan identifies several goals and objectives aimed at addressing this focus, including:

- To encourage transportation planning recognizes the value of sustainable economic development, including tourism and recreation, in the Lakes Region.
- To maintain, enhance, and manage infrastructure that facilitates, encourages, and supports viable pedestrian and bicycling transportation.

*PLAN 2000* then goes further, and identified specific implementation strategies related to bicycle and pedestrian transportation, including:

- Promote alternate modes of transportation as a means of providing congestion and air quality relief, and as an important component of recreation and tourism.
- Provide technical assistance to communities that are planning multi-purpose trails.
- Maintain an inventory of existing facilities.
- Work with towns to develop plans that complement regional and statewide bicycle plans.

#### New Hampshire State Rail Plan (2001)

The 2001 New Hampshire State Rail Plan identifies both the active and the abandoned rail lines in New Hampshire. While there is some discussion of possibly upgrading the rail system in the Lakes Region, no definitive plans are currently available. It is helpful to know the possible uses for the railroad right of way for planning purposes. It is a stated goal in this plan that abandoned railroad rights of way that are owned by the state should be preserved and eventually put to public use such as trails.



#### New Hampshire State Trails Plan (2005)

This plan is intended to; identify existing multi-use trails and corridors with the potential to be developed or improved as multi-use trails, describe the conditions and potential of these trails and corridors for future trail use, and propose guidance for trail development. It identifies and describes abandoned railroad corridors owned by the state, catalogs other non-railroad off-street trails and bike/pedestrianoriented projects throughout the state, summarizes planning studies and documents that address offstreet trails, pedestrian accommodations, and bicycle features, and provides guidelines for developing trails.


## Appendix B:

## Survey Questions and Results

This appendix contains the results of the online survey that was conducted in the winter of 2022, to which 634 people responded. The results are displayed with charts. The vertical axis often indicates the *number* of responses to the question, not the percentage. The horizontal axis contains the set of answers to each question. Not all Respondents answered all questions. Many questions allowed for multiple answers.



What is your age?







What are the top 3 reasons that you bike?







Where do you primarily walk?

Do you ever ride a bicycle?







Where do you primarily bike?





What prevents you from walking more or at all?









What would encourage you to walk more?

What would encourage you to bike more?







In your community, are there places where new or repaired sidewalks are needed?



In your city/town, would you like more off-street paths / trails?









## Appendix C:

Local Master Plan Goals and Recommendations Related to Non-Motorized Transportation

#### Alexandria: 2014

Alexandria's master plan contains a small section on recreation in the community. The stated goal is to increase and expand the types of recreation available. The recommendations to achieve this are to assemble existing recreational groups to attend yearly advisory meetings, develop trails for non-motorized use, increase activities and events, and develop a recreation field in the town center.

#### Andover: 2013

Andover's master plan does not mention pedestrian and bicycling facilities. Continuing the 1992 plan, it states that the community is open to an increased recreation and tourism economy. The recreation project mentioned that most closely relates to pedestrian facilities is the mentioning of a "pick-your-own-produce" farm that would encourage walking throughout the farm.

#### Ashland: 2011

Ashland's master plan does not contain a pedestrian and bicycling facilities section. However, there were several recommendations related to pedestrian infrastructure in the transportation section of the plan. These included supporting the maintenance of pedestrian and bike pathways, supporting alternative modes of travel in its transportation network, continuing the sidewalk improvement program, and exploring sidewalk expansion to the ball field, US Route 3 to River Street, and along Main Street to West St.

#### Barnstead: 2014

Unlike the 2003 document, Barnstead's 2014 master plan does not mention goals specific to walking and bicycling. Rather, it contains goals regarding the support and preservation of broader recreational opportunities. A survey revealed that 79% of Barnstead residents would like continued support for recreational opportunities, many of which are pedestrian related such as hiking and cross-country skiing. To reflect this, the plan recommends that the Barnstead Conservation Commission and Recreation Commission partner with each other and outdoor groups to support recreation infrastructure by creating a robust 50-year recreation plan with maps included in it.

## Belmont: 2002

Belmont has not updated its master plan since 2002. However, it contains some excellent goals related to pedestrian and bicycle needs. In the recreation chapter, the plan states that the Town wishes to establish, create, expand, and map a series of loop trails to link town lands, scenic and wildlife areas, restaurants, and sleeping/camping facilities throughout the town. This goal was worked toward in 2016 when phase I of III of the Winnisquam Scenic Trail was completed with the help of the Belmont Recreational Alternative Transportation Team (BRATT). The trail begins at Leslie E. Roberts Drive on the Laconia-Belmont line before ending at US Route 3 above Mosquito Bridge. Phase II of this trail is in the planning stages and would further Belmont's recreational goals. In addition, the transportation chapter has the broad goal of facilitating, encouraging, and supporting public transit and non-motorized travel to reduce energy consumption, preserve air quality, and reduce pollution. The key suggestions to achieve this are to support new recreational trails, both motorized and non-motorized, and to create wider shoulders for bike paths along existing and new roads.



#### Bridgewater: 2006

Bridgewater's master plan includes a section specific to bicycle and pedestrian needs. The plan cites serious safety deficiencies in the areas of town where the demand is greatest for walking and biking. These two areas include Whittemore Point, and the area around the school and the town offices along NH Route 3A. The Town of Bridgewater should promote these healthy activities safely through its transportation improvement program. In addition to considering the infrastructure and safety needs of these modes of travel, a special goal of the Town is to provide safe enjoyment not only for walkers, runners, and bicyclists on the roads of the town, but also for hikers, cross-country skiers, and snowmobilers on the back roads and rustic paths of the community. Areas such as the newly established Bridgewater Town Forest along Dick Brown Road and Sawhegenet Park along the Pemigewasset River provide safe recreation areas for a wide array of users.

#### Bristol: 2022

Bristol's master plan is currently under revision. The completed updated chapters outline a focus on maintaining "public safety, recreation, transportation, and the general welfare" while prioritizing "a safe and effective transportation system". These goals require much improved bicycle and pedestrian infrastructure. In its former plan for 2003, Bristol mentioned a lot about revitalizing its village district. The Town stated that "A downtown is, after all, supposed to be a pedestrian area, and the more attractive it is to people on foot, the more successful it is likely to be". This statement gives a good idea of the Town's priorities regarding its improvement intentions. We saw in Chapter 6 that these ambitions were achieved with the recent upgrades to the downtown area. The improvements have proved effective in promoting pedestrian safety in the area. The Town of Bristol has also managed to meet demands by extending the multi-purpose trial and including an entrance in the downtown area.

#### Center Harbor: 2012

The Center Harbor master plan is focused on many things, although the needs of bicyclists and other pedestrians is not one of them. With minimal reference to these needs throughout the plan, Center Harbor's standpoint on bicycle and pedestrian needs is as follows: The town does boast a well-established network of nearly a mile of sidewalks within the village district. Also, the state identifies three roads in Center Harbor on its bicycle map. In addition to these bicycle routes, the Squam Lake Association maintains Belknap Woods and Chamberlain-Reynolds Memorial Forest, which offer off-road walking and biking.

#### Danbury: 2011

Danbury's master plan does not include many references to bicycle and pedestrian needs. The town does have a recreation commission and community center. The Town's master plan does not mention any specific plans for improving wellness or providing new recreational areas beyond what already exists. In 2010, the Town of Danbury conducted a community survey for the purpose of acquiring community input prior to the master plan process. The top two results for the question "Which municipal or community facilities or programs need to be enhanced or improved?" were the "Community Center" at twenty-seven percent (27%) and "Recreational Parks" at twenty-three percent (23%). This indicates that plans for improvements to Danbury's recreational facilities should be made.

## Effingham: 2011

While Effingham's 2003 master plan cited the importance of walking and bicycling to promote physical and mental health and that steps should be taken to promote these activities, this sentiment was not continued into the 2011 update to the master plan. In the 2011 plan, there are transportation and recreation sections, but they do not mention goals or recommendations for walking and bicycling infrastructure; rather, the transportation section regards only motor vehicle transportation.



## Franklin: 2018

When Franklin wrote its former master plan in 2005, bicycle and pedestrian needs were clearly considered. The primary concern of the City relates to the construction and completion of both the Winnipesaukee River Trail and a river walk connecting Trestle View Park to Odell Park. Both projects were accomplished, and Franklin continues its commitment to bike and pedestrian mobility. The 2018 edition focuses on creating recreational areas for outdoor activities including active modes of transportation.

## Freedom: 2020

Freedom's master plan does not contain much about pedestrian and bicycling facilities, however, it does include a paragraph on the current and future infrastructure plans. The plan cites the importance of alternative modes of transportation, yet acknowledges the continued lack of sidewalks in Freedom, impeding safe pedestrian conditions from vehicular traffic. Instead, the best pedestrian facilitation Freedom has existed in the form of four foot wide paved shoulders along Ossipee Lake Road between Babcock Road and Pequawket Trail. There is a proposition to expand the four-foot bike path to Danforth bridge, but no plans exist to extend the bike path the entire length of Ossipee.

## Gilford: 2016

Regarding bicycle and pedestrian needs, Gilford's master plan is brief and focused. The Town identifies that it is not seeing greater use of roadways by pedestrians and bicyclists and that its roads are generally not constructed to safely allow for interaction of vehicles and pedestrians. The Gilford Planning Board supports the development of sidewalks, bicycle paths, walking paths, and similar recreational trails throughout the town. Specifically, the Planning Board encourages these elements to be added throughout the village and around the schools to improve safety in a busy area where many children are present.

#### Gilmanton: 2018

Gilmanton's master plan contains a section on transportation but its plans and recommendations regard strictly motor vehicle transportation rather than pedestrians and bicyclists. However, while there is no mention of it in the 2018 master plan, there is a community trail system in the town that offers recreational opportunities for hikers, bicyclists, equestrians, cross-country skiers, and snowmobilers.

## Hebron: 2010

Hebron's few major roads provide little safety for bicyclists and pedestrians alike. The Town would like to work with NHDOT to make improvements on NH Route 3A, North Shore Road, West Shore Road, and Groton Road to improve pedestrian safety. The updated vision statement focused largely on improving recreational opportunities for active transportation, rather than regional connectivity. The Town would like to add shoulders to their own roads to provide safer travel and recreation for pedestrians and bicyclists alike. They have also made recommendations in the land use chapter to expand recreational facilities on town-owned lands, work with land trusts, and private municipal owners to create linking trail networks, explore the possibility of a Cockermouth River walk, and encourage the maintenance of class VI roads as four-season multi-use trails.

#### Hill: 2007

The Town of Hill is in the "development" stage of its bicycle and pedestrian planning. The master plan makes note that a limited network of sidewalks and streetlights exist in the village and are mostly located within a two-block area of the school. The Town feels that these features, although nice, are not practical on the outlying roads. Some ideas for improvement include seeking grants available to fund sidewalk upgrades within the village, exploring ways to fund the maintenance of its portion of the Heritage Trail, and requiring new subdivisions to have open space and incorporate room for trails. These trails could eventually connect to an existing trail network, saving the town money, and increasing recreational opportunities for the townspeople.









#### Holderness: 2015

The 2015 update to the *Holderness Master Plan Vision* does not mention anything about specific improvements or goals for their transportation system. The former master plan from 2007 includes a bit more about the needed improvements in its bicycle and pedestrian system. The rarely used 50-acre town forest boasts a network of logging roads that can be used for hiking in the summer and cross-country skiing in the winter. The Town attributes the limited use of the town forest to the unfamiliarity of its location. This may be the driving factor behind the Town's goal of creating and distributing a town trail map which would include routes to hike, walk, bike, ski, and snowmobile within Holderness town limits. In addition to the fifty miles of trails maintained by the Squam Lakes Association, the Town would like to develop a town-wide multi-modal trail system.

## Laconia: 2018 Draft

While the City of Laconia has drafted a 2018 master plan, it does not yet contain a transportation section. In its 2007 master plan, Laconia admits to having poor bicycle and pedestrian infrastructure and has been taking steps to correct that. Luckily, many improvements are currently underway or have been completed. The City finished a sidewalk project on Elm Street and finalized a joint effort with the state to provide sidewalks for several thousand feet of Endicott Street on either side of its intersection with Weirs Boulevard. Laconia generally requires sidewalks in new developments. The City has also been actively pursuing the construction of new bicycle and pedestrian facilities throughout the city, including the Court Street Bicycle Path, the Winnipesaukee Opechee Winnisquam (WOW) Trail, and the bicycle and pedestrian bridge on Bridge Street. The city network will also become part of the Lakes Region Bikeway, which is proposed to stretch from Meredith to Franklin. The Laconia Trails with Rails Exploratory Committee has been a driving force on these issues. To incorporate their activities into a city-wide trails network, Laconia's strategy is to invest in non-motorized transportation systems such as the riverwalk, sidewalks, and the Rails with Trails program.

Additionally, the City of Laconia provides an interactive walking guide created with Google Maps found on the City's website under the Visitors tab > Maps > Laconia Walking Map.

#### Meredith: 2002

The Meredith Planning Board is currently in the process of updating its master plan. In the 2002 plan, Meredith makes its bicycle and pedestrian needs a top priority. So serious is it that the master plan makes these transportation needs priority number two out of six. Some of the Town's objectives in dealing with these needs are to develop a bicycle and pedestrian plan. This plan will: (1) Include an inventory of all existing pedestrian and bicycle facilities. (2) Include an inventory of activity centers or neighborhoods that ought to be connected. (3) Identify specific areas where networks can be expanded or improved. (4) Identify needed support facilities (i.e., restrooms, drinking water, bicycle racks, benches, parking areas, etc.) (5) Include an education and awareness component to encourage greater utilization and to reduce the exposure to the risks associated with these forms of travel. The plan should be considered in the scoping of all public and private development projects. The Town hopes to provide a network of multi-purpose facilities to improve travel opportunities to, from, and within Meredith. Lastly, the Town hopes to incorporate bicycle and pedestrian access into land use regulatory processes.

Additionally, the Meredith Pathway Committee is an advisory committee appointed by the Meredith Board of Selectmen. The broadly stated purpose of the committee is to develop a pedestrian plan and promote pedestrian activity in the community. The committee provides a pathway map of designated routes throughout the town.



## Moultonborough: 2016

Much of the work mentioned in the master plan is derived from Moultonborough's 2010 Travel Plan. The plan to revitalize its village includes many suggestions to update bicycle and pedestrian systems. One of the major improvements is building sidewalks on NH Route 25 in the village, which is currently in the engineering phase. In addition, Moultonborough completed Phase II of the Moultonborough Neck Pathway in 2007. Phase II runs along Moultonborough Neck Road to Kona Farms Road. Phase I runs from JoJo's Country Store to Winaukee Road, leaving the middle of the neck for a Phase III. The construction of this pathway has encouraged many residents to promote alternative modes of transportation and increased the demand for bike and pedestrian connectivity throughout the community.

## New Hampton: 2021

Regarding bicycle and pedestrian needs, New Hampton's 2021 update to the master plan mentions very little. One of their main focuses for transportation is the high volume and speed of its main corridors including NH Route 104 and NH Route 132. Their recommendations to improve safety in these areas are to implement traffic calming measures. These elements would benefit pedestrians and vulnerable road users as well as drivers. Unlike the previous plan from 2002, this one does not emphasize the need for recreational areas in relation to the transportation system and multi-modal transportation.

## Northfield: 2014

The Northfield master plan briefly touches on several topics related to bicycle and pedestrian needs. These topics include a mixed-use village district like the plan of nearby New Hampton. The update mentions that a sidewalk issue was addressed on Elm Street to improve pedestrian safety. Contained within Northfield are two excellent recreational areas: The Winnipesaukee River Trail, which has been extended since the last master plan, and Knowles Pond Park. Together they offer a network of trails that allow access by foot, ski, peddle, snowshoe, and equestrian use. The Town would like a defined bicycle and pedestrian linkage to connect and enhance recreational facilities for people of all ages and levels of ability. They have hope that the "Missing Link" bridge over the Winnipesaukee River will be funded in the future to complete the multi-use trail.

## Ossipee: 2010

Ossipee does consider bicyclists and pedestrians in its master plan. The Town plans to install a bicycle path around Ossipee Lake that will connect with existing pathways around the lake and eventually to non-motorized pathways in the rest of the Lakes Region. The regional Transportation Advisory Committee has been instrumental in the development of these bicycle paths throughout the Lakes Region. The Ossipee Conservation Commission plans to develop trails for hiking on conservation land in the town. One of the first properties they plan to work on is Sumner Brook. Constitution Park already contains a trail system including wooden bog bridges to prevent adverse impact from foot traffic. Lastly, the master plan recommends that sidewalks and crosswalks be considered anywhere pedestrian traffic exists within the town.

## Plymouth: 2018

The Town of Plymouth has many recommendations related to bicycling and pedestrian infrastructure outlined in the implementation section of its master plan. First, the Town wants to develop a plan for safe pedestrian and bicycle access and circulation in addition to the comprehensive master plan. This



would include development guidelines for incremental improvement in the appearance of the area and in pedestrian/bicycle safety. To achieve this, Plymouth wants to seek funding from NHDOT's Transportation Alternatives Program to pay for improvements to pedestrian and bicycling infrastructure.

## Sanbornton: 2012

Sanbornton's master plan mentions mostly motor vehicle planning. However, there are two recommendation points in it about pedestrian and bicycling infrastructure. The plan recommends that public right of way be accessible to alternative modes of transportation including pedestrians, bicyclists, and equestrians. It also points out that the infrastructure should be designed to include areas for additional utilities where necessary. Lastly, it mentions that long-range transportation policies should encourage transportation alternatives such as car-pooling, bicycling, and other energy-efficient transportation modes.

#### Sandwich: 2011

Sandwich's master plan acknowledges several modes of transportation including bicycling and pedestrian facilities. The recent update admits that Town roadway facilities for bicyclists and pedestrians are limited, especially in Center Sandwich. The plan recommends potential actions that could be taken to achieve overall bicycle and pedestrian facility improvement goals. These actions include: (1) Adding an additional 2-feet of pavement to the shoulder of roads to provide a safe place for bicyclists and pedestrians; (2) Working with NHDOT to assure that any state bridges that are rebuilt or reconstructed provide adequate space for sidewalks and/or bicycle lanes; (3) Develop a long-range plan for sidewalks in Center Sandwich; (4) Properly mark and sign the state designated bicycle routes, including NH Routes 113 and NH Route 109; (5) Work toward a system of bicycle routes and multi-use trails/paths for the enjoyment of Sandwich citizens and visitors that is coordinated with state and regional trail systems.

#### Tamworth: 2008

Unfortunately, in some areas the historic rural nature of the town left little physical room for improvements to transportation systems. Regardless, improvements including the installation of a crosswalk in Chocorua Village were in the plans for 2008. This improvement supports Tamworth's second smart growth principle 'Foster the traditional character of New Hampshire downtowns, villages, and neighborhoods by encouraging a human scale of development that is comfortable for pedestrians and conducive to community life.' Community survey respondents indicated a high degree of support for more bicycle paths, sidewalks, and linked trails throughout the town. Development of town bicycle lanes is on the forefront of the 2008 Plan. A local resident is spearheading a project to improve the shoulders on both sides of NH Route 113. Eighty percent (80%) of the project is covered by federal funds while the other twenty percent (20%) is projected to cost the town \$29,000. These improvements will make the roadway safer for pedestrians and bicyclists.

#### Tilton: 2013

Tilton's master plan cites the need for open space and recreation areas to provide an improved quality of life, sense of town character, and economic boost. An example of this is the walking trail from NH Route 140 to near downtown Tilton developed by the Winnipesaukee River Trail Association. The Town's goal is to extend the trail to downtown and eventually cross the Merrimack River to connect with Northfield's portion of the trail. This walking trail is the only pedestrian infrastructure mentioned in the plan. While there is a section on alternative modes of transportation, its recommendations are motor vehicle related like bus network expansion and road improvements.



## Tuftonboro: 2021

Tuftonboro's 2021 community attitude survey indicated great support for improving bicycle and pedestrian access throughout the town. To please its residents, the master plan outlines many goals and recommendations related to improving pedestrian and bicycling facilities. It is recommended that Tuftonboro petitions the state to reduce highway lane width from 12' to 10' to accommodate 4' bicycle lanes in each direction, improving safety and eliminating the perception that the roads are not bicycle-friendly. Additionally, the Town wants to add curb bump-outs and table intersections to further improve pedestrian safety.

## Wolfeboro: 2019

Wolfeboro maintains 7.5 miles of sidewalk and 1,750 feet of stone dust path within the town. Since the 2007 master plan was written, pedestrian movement in the downtown has increased (anecdotally) after significant upgrades were completed in the downtown area. The extension of sidewalks along Bay Street, Center Street, North Main Street, and Pine Hill Road are in the planning stages. Despite this relatively robust sidewalk network, Wolfeboro concedes that there are several safety issues around them including minimal separation from travel lanes, multiple and wide curb cuts, and accessibility constraints.

For improvements to this infrastructure, the master plan outlines many general and sidewalk-specific recommendations. It suggests that a bicycle, pathway, and pedestrian plan/map be developed to supplement the master plan. Plus, it recommends implementing traffic calming measures, improving crosswalk locations and designs in downtown, and encouraging the development of a circular back bay pedestrian sidewalk and pathway system which would connect three of the four commercial sub areas.

Specific to sidewalks, it is suggested that a sidewalk/pedestrian pathway inventory and asset management plan be developed. Expansions to sidewalks on North Main Street, Bay Street, Center Street to Mast Landing, Pine Street, and the Elm Street Pathway are also suggested. Lastly, where sidewalk construction is not feasible, the plan recommends that pedestrian safety still be improved via speed limit reduction, travel lane narrowing, provision of sharrows, and shoulder area expansion.



## Appendix D:





\*\*

56





Attachment 5

# **CERTIFICATE OF AUTHORITY**

I, Lianne Prentice, do hereby certify that:

- 1. I am the duly elected Secretary of the Lakes Region Planning Commission, a regional planning agency established pursuant to the laws of the State of New Hampshire (RSA 36:45-53).
- 2. I sign and maintain, or cause to be maintained, and am familiar with the minutes of Commission.
- 3. I am duly authorized to issue certificates with respect to the contents of such minutes.
- 4. At its regular meeting held on June 12, 2024, the Executive Board of the Commission voted to grant the Commission's Executive Director, Shanna B. Saunders, the authority to apply for funding opportunities, accept monies, and execute any documents which may be necessary to effectuate contracts. This authorization remains in full force and effect until it is revoked.
- 5. This authorization has not been revoked, annulled or amended in any manner whatsoever, and remains in full force and effect as of the date hereof.
- 6. The following person has been appointed to and now occupies the office indicated in (4) above:

## Shanna B. Saunders, Executive Director

IN WITNESS WHEREOF, I have hereunto set my hand as the Secretary of the Lakes Region Planning Commission, this the \_\_\_\_\_ day of \_\_\_\_\_, 2024.

Lianne Prentice, Secretary

## STATE OF NEW HAMPSHIRE COUNTY OF BELKNAP

On this the \_\_\_\_\_ day of \_\_\_\_\_, 2024, before me, \_\_\_\_\_, the undersigned officer, personally appeared Lianne Prentice who acknowledged herself to be the Secretary of the Lakes Region Planning Commission and, being authorized to do so, executed the foregoing instrument for the purpose therein contained.

IN WITNESS WHEREOF, I have set my hand and official seal.

Notary Public	
My Commission Expires:	